

Time-varying monetary-policy rules and financial stress

Baxa, Jaromír; Horváth, Roman; Vašíček, Bořek 2011 Dostupný z http://www.nusl.cz/ntk/nusl-124042

Dílo je chráněno podle autorského zákona č. 121/2000 Sb.

Tento dokument byl stažen z Národního úložiště šedé literatury (NUŠL). Datum stažení: 26.05.2024

Další dokumenty můžete najít prostřednictvím vyhledávacího rozhraní nusl.cz .

WORKING PAPER SERIES 3

Jaromír Baxa, Roman Horváth, Bořek Vašíček: Time-Varying Monetary-Policy Rules and Financial Stress: Does Financial Instability Matter for Monetary Policy?



WORKING PAPER SERIES

Time-Varying Monetary-Policy Rules and Financial Stress: Does Financial Instability Matter for Monetary Policy?

> Jaromír Baxa Roman Horváth Bořek Vašíček

> > 3/2011

CNB WORKING PAPER SERIES

The Working Paper Series of the Czech National Bank (CNB) is intended to disseminate the results of the CNB's research projects as well as the other research activities of both the staff of the CNB and collaborating outside contributor, including invited speakers. The Series aims to present original research contributions relevant to central banks. It is refereed internationally. The referee process is managed by the CNB Research Department. The working papers are circulated to stimulate discussion. The views expressed are those of the authors and do not necessarily reflect the official views of the CNB.

Distributed by the Czech National Bank. Available at http://www.cnb.cz.

Reviewed by: Sofia Bauducco (Central Bank of Chile) Aleš Bulíř (International Monetary Fund) Dana Hájková (Czech National Bank)

Project Coordinator: Michal Franta

© Czech National Bank, July 2011 Jaromír Baxa, Roman Horváth, Bořek Vašíček

Time-Varying Monetary-Policy Rules and Financial Stress: Does Financial Instability Matter for Monetary Policy?

Jaromír Baxa, Roman Horváth and Bořek Vašíček*

We examine whether and how selected central banks responded to episodes of financial stress over the last three decades. We employ a new monetary-policy rule estimation methodology which allows for time-varying response coefficients and corrects for endogeneity. This flexible framework applied to the USA, the UK, Australia, Canada, and Sweden, together with a new financial stress dataset developed by the International Monetary Fund, not only allows testing of whether central banks responded to financial stress, but also detects the periods and types of stress that were the most worrying for monetary authorities and quantifies the intensity of the policy response. Our findings suggest that central banks often change policy rates, mainly decreasing them in the face of high financial stress. However, the size of the policy response varies substantially over time as well as across countries, with the 2008–2009 financial crisis being the period of the most severe and generalized response. With regard to the specific components of financial stress, most central banks seemed to respond to stock-market stress and bank stress, while exchange-rate stress is found to drive the reaction of central banks only in more open economies.

JEL Codes: E43, E52, E58.

Keywords: Endogenous regressors, financial stress, monetary policy, Taylor rule, time-varying parameter model.

^{*} Jaromír Baxa, Institute of Economic Studies, Charles University, Prague and Institute of Information Theory and Automation, Academy of Sciences of the Czech Republic (e-mail: jaromir.baxa@gmail.com)

Roman Horváth, Institute of Economic Studies, Charles University, Prague (e-mail: roman.horvath@gmail.com) Bořek Vašíček, Czech National Bank (e-mail: borek.vasicek@gmail.com)

We thank Aleš Bulíř, Sofia Bauducco, Øyvind Eitrheim, Dana Hájková, Bernhard Herz, Ekkehart Schlicht, Miloslav Vošvrda, and seminar participants at the 7th Norges Annual Monetary Policy Conference, the 15th International Conference on Macroeconomic Analysis and International Finance (Rethymno, Greece), the Czech National Bank, the Institute of Information Theory and Automation (Academy of Sciences of the Czech Republic), Universitat de Barcelona, Universitat de Girona, Universitat de les Illes Balears, and Universidad Computense de Madrid for helpful discussions. The views expressed in this paper are not necessarily those of the Czech National Bank. This research was supported by the Grant Agency of the Czech Republic, project no. P402/10/0448.

Nontechnical Summary

The recent financial crisis has intensified the interest in exploring in greater detail the nexus between monetary policy and financial stability. Although keeping the financial system stable is a major task, often delegated to central banks, how to consider financial stability concerns for monetary policy decision-making remains a puzzling question. Monetary policy is likely to react to financial instability in a non-linear way. When a financial system is stable, the interest-rate-setting process largely reflects macroeconomic conditions, and financial stability considerations enter monetary policy discussions only to a limited degree. On the other hand, central banks may alter their monetary policies to reduce financial imbalances if these become severe.

This paper examines the reactions of the main central banks (the US Fed, the Bank of England, the Reserve Bank of Australia, the Bank of Canada, and Sveriges Riksbank) during periods of financial stress over the last three decades. In particular, we estimate the time-varying policy rule of each bank to assess whether and how its policy rate was adjusted in the face of financial instability. We track financial stress by means of a continuous financial stress indicator developed recently by the International Monetary Fund as well as its main subcomponents (banking stress, stock-market stress, and exchange-rate stress). Therefore, our empirical framework is suitable for detecting the periods and types of stress that were perceived as the most worrying and for quantifying the intensity of the policy response.

Although theoretical studies disagree about the viability of considering financial instability for interest-rate setting, our empirical results suggest that central banks often alter the course of monetary policy in the face of high financial stress, mainly by decreasing their policy rates. Yet the size of this response varies substantially over time and across countries as well as in terms of specific types of financial stress. The recent financial crisis evoked the most generalized response and interest rates decreased in the range of 50 to 200 basis points owing solely to financial instability concerns (above what could be attributed to the decline in inflation expectations and output below its potential). However, significant interest rate adjustment to financial instability has been found for all central banks over the whole sample period. In terms of the specific components of stress, most banks seemed to respond to stock-market stress and bank stress, while exchange-rate stress drove central bank reactions only in more open economies.

The results also point to the usefulness of augmenting the standard version of the monetary policy rule by some measure of the financial conditions to get a better understanding of the interest-rate-setting process, especially when financial markets are not stable.

1. Introduction

The recent financial crisis has intensified the interest in exploring the interactions between monetary policy and financial stability. Official interest rates were driven sharply to historical lows, and many unconventional measures were used to pump liquidity into the international financial system. Central banks pursued monetary policy under high economic uncertainty coupled with large financial shocks in many countries. The financial crisis also raised new challenges for central bank policies, in particular the operationalization of issues related to financial stability for monetary-policy decision making (Goodhart, 2006; Borio and Drehmann, 2009).

This paper seeks to analyze whether and how central banks reacted to periods of financial instability and, in particular, whether and how the interest-rate-setting process evolved in response to financial instability over the last three decades. The monetary policies of central banks are likely to react to financial instability in a non-linear way (Goodhart et al., 2009). When a financial system is stable, the interest-rate-setting process largely reflects macroeconomic conditions, and financial stability considerations enter monetary policy discussions only to a limited degree. On the other hand, central banks may alter their monetary policies to reduce financial imbalances if these become severe. In this respect, Mishkin (2009) questions the traditional linear-quadratic framework¹ when financial markets are disrupted and puts forward an argument for replacing it with non-linear dynamics describing the economy and a non-quadratic objective function resulting in non-linear optimal policy.

To address the complexity of the nexus between monetary policy and financial stability as well as to evaluate monetary policy in a systematic manner, this paper employs the recently developed time-varying parameter estimation of monetary-policy rules, appropriately accounting for endogeneity in policy rules. This flexible framework, together with a new comprehensive financial stress dataset developed by the International Monetary Fund, will allow not only testing of whether central banks responded to financial stress, but also quantification of the magnitude of this response and detection of the periods and types of stress that were the most worrying for monetary authorities.

Although theoretical studies disagree about the role of financial instability for central banks' interest-rate-setting policies, our empirical estimates of the time-varying monetary-policy rules of the US Fed, the Bank of England (BoE), the Reserve Bank of Australia (RBA), the Bank of Canada (BoC), and Sveriges Riksbank (SR) show that central banks often alter the course of monetary policy in the face of high financial stress, mainly by decreasing policy rates.² However, the size of this response varies substantially over time as well as across countries. There is some cross-country and time heterogeneity as well when we examine central banks' considerations of

¹ That is, linear behavior of the economy and a quadratic objective function of the monetary authority.

 $^{^2}$ Our choice of countries is based on data availability and on the suitability of the data for our econometric framework. Due to limited data availability, we do not include the Reserve Bank of New Zealand, the ECB, and emerging countries. The Bank of Japan could not be included either, given that its policy rates were flat for an extended period.

specific types of financial stress: most of them seemed to respond to stock-market stress and bank stress, and exchange-rate stress drives central bank reactions only in more open economies.

The paper is organized as follows. Section 2 discusses related literature. Section 3 describes our data and empirical methodology. Section 4 presents our results. Section 5 concludes. An appendix with a detailed description of the methodology and additional results follows.

2. Related Literature

First, this section gives a brief overview of the theory as well as empirical evidence on the relationship between monetary policy (rules) and financial instability. Second, it provides a short summary of various measures of financial stress.

2.1 Monetary Policy (Rules) and Financial Instability – Some Theories

Financial friction, such as unequal access to credit or debt collateralization, is recognized as having important consequences for monetary policy transmission, and Fisher (1933) has already presented the idea that adverse credit-market conditions can cause significant macroeconomic disequilibria.

During the last two decades, the effects of monetary policy have been studied mainly within New Keynesian (NK) dynamic stochastic general equilibrium (DSGE) models, which assume the existence of nominal rigidities. The common approach to incorporating financial market friction within the DSGE framework is to introduce the financial accelerator mechanism (Bernanke et al., 1996, 1999), implying that endogenous developments in credit markets work to amplify and propagate shocks to the macro economy. Tovar (2009) emphasizes that the major weakness of the financial accelerator mechanism is that it only addresses one of many possible financial frictions. Goodhart et al. (2009) note that many NK DSGE models lack the financial sector completely or model it in a rather embryonic way. Consequently, more recent contributions within this stream of literature have examined other aspects of financial friction, such as balance sheets in the banking sector (Choi and Cook, 2004), the portfolio-choice issue with complete (Engel and Matsumoto, 2009) or incomplete markets (Devereux and Sutherland, 2007), and collateral constraints (Iacovello and Neri, 2010).³

A few studies focus more specifically on the relationship between the monetary-policy stance (or the monetary-policy rule) and financial stability. However, they do not arrive at a unanimous view of whether a monetary-policy rule should include some measure of financial stability. Brousseau and Detken (2001) present an NK model where a conflict arises between short-term price stability and financial stability due to a self-fulfilling belief linking the stability of inflation to the smoothness of the interest-rate path and suggests that monetary policy should react to financial instability. Akram et al. (2007) investigate the macroeconomic implications of pursuing

³ A survey of this literature is provided by Tovar (2009).

financial stability within a flexible inflation-targeting framework. Their model, using a policy rule augmented by financial-stability indicators, shows that the gains of such an augmented rule vis-à-vis the rule without financial-stability indicators highly depends on the nature of the shocks. Akram and Eitrheim (2009) build on the previous framework, finding some evidence that the policy response to housing prices, equity prices or credit growth can cause high interest-rate volatility and actually lower financial stability in terms of indicators that are sensitive to interest rates. Cecchetti and Li (2008) show, in both a static and dynamic setting, that a potential conflict between monetary policy and financial supervision can be avoided if the interest-rate rule takes into account (procyclical) capital-adequacy requirements, in particular, that policy interest rates are lowered when financial stress is high. Bauducco et al. (2008) extend the current benchmark NK model to include financial systems and firms that require external financing. Their simulations show that if a central bank responds to financial instability by policy easing, it achieves better inflation and output stabilization in the short term at the cost of greater inflation and output volatility in the long term, and vice versa. For the US Fed, Taylor (2008) proposes a modification of the standard Taylor rule to incorporate adjustments to credit spreads. Teranishi (2009) derives a Taylor rule augmented by the response to credit spreads as an optimal policy under heterogeneous loan-interest-rate contracts. He finds that the policy response to a credit spread can be both positive and negative, depending on the financial structure. However, he also proposes that when nominal policy rates are close to zero, a commitment rather than a discretional policy response is the key to reducing credit spreads. Christiano et al. (2008) suggest augmenting the Taylor rule with aggregate private credit and find that such a policy would raise welfare by reducing the magnitude of the output fluctuations. Cúrdia and Woodford (2010) develop an NK DSGE model with credit friction to evaluate the performance of alternative policy rules that are augmented by a response to credit spreads and to aggregate the volume of private credit in the face of different shocks. They argue that the response to credit spreads can be welfare improving, but the optimal size of such a response is probably rather small. Like Teranishi (2009), they find little support for augmenting the Taylor rule by the credit volume, given that the size and even the sign of the desired response is sensitive to the sources of shock and their persistence, which is information that is not always available during operational policy making.

A related stream of literature focuses on the somewhat narrower issue of whether or not monetary policy should respond to asset prices. Bernanke and Gertler (1999, 2001) argue that the stabilization of inflation and output provides a substantial contribution to financial stability and that there are few, if any, gains to responding to asset prices. Faia and Monacelli (2007) extend the model developed by Bernanke and Gertler (2001) by a robust welfare metric, confirming that strict inflation stabilization offers the best solution. Cecchetti et al. (2000) take the opposite stance, arguing that developments in asset markets can have a significant impact on both inflation and real economic activity, and central banks might achieve better outcomes by considering asset prices provided they are able to detect asset-price misalignments. Borio and Lowe (2002) support this view, claiming that financial imbalances can build up even in a low-inflation environment, which is normally favorable to financial stability. The side effect of low inflation is that excess demand pressures may first appear in credit aggregates and asset prices rather than consumer prices, which are normally considered by policy makers. Gruen et al. (2005) argue that responding to an asset bubble is feasible only when the monetary authority is able to make a

correct judgment about the process driving the bubble. Roubini (2006) and Posen (2006) provide a summary of this debate from a policy perspective.

2.2 Monetary Policy (Rules) and Financial Instability – Empirical Evidence

The empirical evidence on central banks' reactions to financial instability is rather scant. Following the ongoing debate about whether central banks should respond to asset-price volatility (e.g. Bernanke and Gertler, 1999, 2001; Cecchetti et al., 2000; Bordo and Jeanne, 2002), some studies have tested the response of monetary policy to different asset prices, most commonly stock prices (Rigobon and Sack, 2003; Chadha et al., 2004; Siklos and Bohl, 2008; Fuhrer and Tootell, 2008). They find some evidence either that asset prices entered the policy-information set (because they contain information about future inflation) or that some central banks were directly trying to offset these disequilibria.⁴ All of these papers estimate time-invariant policy rules, which means that they test a permanent response to these variables. However, it seems more plausible that if central banks respond to asset prices, they do so only when asset-price misalignments are substantial; in other words, their responses are asymmetric. There are two additional controversies related to the effects of asset prices on monetary-policy decisions. The first concerns the measure, in particular whether the stock-market index that is typically employed is sufficiently representative, or whether some other assets, in particular housing prices, should be considered as well. The second issue is related to the (even ex-post) identification of asset-price misalignment. Finally, it is likely that the perception of misalignments is influenced by general economic conditions and that a possible response might evolve over time.

Detken and Smets (2004) summarize some stylized facts on macroeconomic and monetary-policy developments during asset-price booms. Overall, they find that monetary policy was significantly looser during high-cost booms that were marked by crashes of investment and real-estate prices in the post-boom periods.

A few empirical studies measure the monetary-policy response using broader measures of financial imbalances. Borio and Lowe (2004) estimate the response of four central banks (the Reserve Bank of Australia, the Bundesbank, the Bank of Japan, and the US Fed) to imbalances proxied by the ratio of private-sector credit to GDP, inflation-adjusted equity prices, and their composite. They find either negative or ambiguous evidence for all countries except the USA, confirming that the Fed responded to financial imbalances in an asymmetric and reactive way, i.e., that the federal funds rate was disproportionately lowered in the face of imbalance unwinding, but was not tightened beyond normal as imbalances built up. Cecchetti and Li (2008) estimate a Taylor rule augmented by a measure of banking stress, in particular the deviation of leverage ratios (total loans to the sum of equity and subordinated debt; total assets to the sum of bank capital and reserves) from their Hodrick-Prescott trend. They find some evidence that the Fed adjusted the interest rate to counteract the procyclical impact of a bank's capital requirements, while the Bundesbank and the Bank of Japan did not. Bulíř and Čihák (2008) estimate the monetary-policy response to seven alternative measures of financial-sector vulnerability (crisis probability, time to crisis, distance to default or credit default swap spreads)

⁴ A similar but somewhat less polemic debate applies to the role of exchange rates, especially for small, open economies (Taylor, 2001).

in a panel of 28 countries. Their empirical framework is different in the sense that the monetarypolicy stance is proxied along the short-term interest rate by measures of domestic liquidity, and external shocks are controlled for. In the panel setting, they find a statistically significant negative response to many variables representing vulnerability (policy easing) but, surprisingly, not in country-level regressions. Belke and Klose (2010) investigate the factors behind the interest-rate decisions of the ECB and the Fed during the current crisis. They conclude that the estimated policy rule was significantly altered only for the Fed, and they put forward that the ECB gave greater weight to inflation stabilization at the cost of some output loss.

2.3 Measures of Financial Stress

The incidence and determinants of different types of crises have been typically traced in the literature by a means of narrative evidence (expert judgment). This has sometimes been complemented by selected indicators (exchange rate devaluation or the state of foreign reserves) that point to historical regularities (e.g., Eichengreen and Bordo, 2002; Kaminsky and Reinhart, 1999; Reinhart and Rogoff, 2008; Laeven and Valencia, 2008). The empirical studies (e.g., Goldstein et al., 2000) used binary variables that were constructed based on these narratives.

Consequently, some contributions strived to provide more data-driven measures of financial stress. Most of the existing stress indices are based on high-frequency data, but they differ in the selected variables (bank capitalization, credit ratings, credit growth, interest rate spreads or volatility of different asset classes), country coverage, and the aggregation method. An important advantage of continuous stress indicators is that they may reveal periods of small-scale stress that did not result in full-blown crises and were neglected in studies based on binary crisis variables.

The Bank Credit Analyst (BCA) reports a monthly financial stress index (FSI) for the USA that is based on the performance of banking shares compared to the whole stock market, credit spreads and the slope of the yield curve, and new issues of stocks and bonds and consumer confidence. JP Morgan calculates a Liquidity, Credit and Volatility Index (LCVI) based on seven variables: the US Treasury curve error (the standard deviation of the spread between on-the-run and off-the-run US Treasury bills and bonds along the entire maturity curve), the 10-year US swap spread, US high-yield spreads, JP Morgan's Emerging Markets Bond Index, foreign exchange volatility (the weighted average of the 12-month implied volatilities of several currencies), the Chicago Board of Exchange VIX equity volatility index, and the JP Morgan Global Risk Appetite Index.

Illing and Liu (2006) develop a comprehensive FSI for Canada. Their underlying data cover equity, bond, and foreign exchange markets as well as the banking sector. They use a standard measure and refined measure of each stress component, where the former refers to the variables and their transformations that are commonly found in the literature, while the latter incorporates adjustments that allow for better extraction of information about stressful periods. They explore different weighting schemes to aggregate the individual series (factor analysis, the size of the corresponding market for total credit in the economy, variance-equal weighting). Finally, they perform an expert survey to identify periods that were perceived as especially stressful, confirming that the FSI matches these episodes very well.

For the Fed Board of Governors, Carlson et al. (2008) propose a framework similar to the optionpricing model (Merton, 1974) that aims to provide the distance-to-default of the financial system, the so-called Index of Financial Health. The method uses the difference between the market value of a firm's assets and liabilities and the volatility of the asset's value to measure the proximity of a firm's assets to being exceeded by their liabilities. They apply this measure to 25 of the largest US financial institutions, confirming its impact on capital investments in the US economy. The Kansas City Fed developed the Kansas City Financial Stress Index (Hakkio and Keeton, 2009), which is published monthly and is based on eleven variables (seven spreads between different bond classes by issuers, risk profiles and maturities, correlations between returns on stocks and Treasury bonds, expected volatility of overall stock prices, volatility of bank stock prices, and a cross-section dispersion of bank stock returns) that are aggregated by principal component analysis.

Finally, the International Monetary Fund (IMF) recently published financial stress indices for various countries. Cardarelli et al. (2011) propose a comprehensive index based on high-frequency data where the price changes are measured with respect to their previous levels or trend values. The underlying variables are standardized and aggregated into a single index (FSI) using variance-equal weighting for each country and period. The FSI has three subcomponents: the banking sector (the slope of the yield curve, TED spread, and the beta of banking-sector stocks), securities markets (corporate bond spreads, stock-market returns and time-varying volatility of stock returns) and exchange rates (time-varying volatility of NEER changes). Balakrishnan et al. (2009) modify the previous index to account for the specific conditions of emerging economies, on the one hand including a measure of exchange rate pressures (currency depreciation and decline in foreign reserves) and sovereign debt spread, and on the other hand downplaying the banking-sector measures (slope of the yield curve and TED spread).⁵ We will use the former index, given its comprehensiveness as well as its availability for different countries (see more details below).

3. Data and Empirical Methodology

3.1 The Dataset

Given the frequency of monetary policy committee meetings in most central banks, we use monthly data (due to unavailability of all monthly series for a sufficiently long time period, we use quarterly data for Sweden and Canada). The sample periods vary slightly due to data availability (the US 1981:1M–2009:6M; the UK 1981:1M–2009:3M; Australia 1983:3M–2009:5M; Canada 1981:1Q–2008:4Q; Sweden 1984:2Q–2009:1Q).

The dependent variable is typically an interest rate closely related to the official (censored) policy rate, in particular the federal funds rate (3M) for the USA, the discount rate (three-month Treasury bills) for the UK, Canada, and Sweden, and the three-month RBA-accepted bills rate for Australia. It is evident that the policy rate is not necessarily the only instrument that central banks use, especially during the 2008–2009 global financial crisis, when many unconventional measures were implemented (see Borio and Disyatat, 2009; Reis, 2010). To address this issue in

⁵ The IMF Financial Stress Index has recently been applied by Melvin and Taylor (2009) to analyze exchange rate crises.

terms of estimated policy rules, for a robustness check we use the interbank interest rate (at a maturity of three months). While both rates are used in empirical papers on monetary-policy rule estimation without great controversy, the selection of the interest rate becomes a more delicate issue during periods of financial stress (Taylor, 2008). While the former is more directly affected by genuine monetary-policy decisions (carried out by open market operations), the latter additionally includes liquidity conditions on interbank markets and, as such, can be affected by unconventional policies, though these are usually insulated (often intentionally) from policy interest rates.⁶ This is a drawback but also a potential advantage of this alternative dependent variable.⁷ On the one hand, changes in official policy rates may not pass through fully to interbank interest rates, in particular when the perceived counterparty risk is too high and credit spreads widen (see Taylor and Williams, 2009). On the other hand, the interbank rate may also incorporate the impact of policy actions, such as quantitative easing aimed at supplying additional liquidity into the system.⁸

Inflation is measured as the year-on-year change in the CPI, apart from for the United States, where we use the personal consumption expenditures price index (PCE), and Sweden, where underlying CPIX inflation (which excludes households' mortgage-interest expenditures and the direct effects of changes in indirect taxes and subsidies from the CPI) is used.⁹ The output gap is proxied by the gap of the seasonally adjusted industrial production index derived by the Hodrick-Prescott filter with a smoothing parameter set to 14,400.¹⁰ For Sweden and Canada, where we use quarterly data, the output gap was taken as reported in the OECD Economic Outlook (production function method based on NAWRU — non-accelerating wage rate of unemployment).

⁶ Borio and Disyatat (2009) characterize unconventional policies as policies that affect the central bank's balance sheet size and composition and that can be insulated from interest rate policy (the so-called "decoupling principle"). One common example of such a policy (not necessarily used during times of crisis) is sterilized exchange-rate intervention. Given that we are looking not at a single episode of stress, but rather want to identify whether monetary authorities deviated from systematic patterns (the policy rule) during these periods (by responding to indicators of financial stress), we need to use a consistent measure of policy action that is adjusted during periods of financial stress, though other measures may be in place as well. Therefore, we assume that the monetary-policy stance is fully reflected in the interest rate, and we are aware that it might be subject to downward bias on the financial-stress coefficient. The reader may want to interpret our results on the importance of financial stress for interest-rate setting as a conservative estimate.

⁷ The experience from the recent crisis represents an evident drawback of using interbank rates as a proxy for monetary policy intentions. In particular, the interbank markets froze at longer maturities (3–12 months) and the term premium increased sharply in most countries. In addition, in open economies, central banks lost control over very short maturities. For example, EONIA has been below the ECB repo rate for an extended period as major financial institutions can borrow more cheaply in the yen or dollar markets.

⁸ There are other policy measures that can be used as a reactive or pre-emptive response to financial stress, such as regulatory or administrative measures, although their effects are likely to appear only in the longer term and cannot be reasonably included in our empirical analysis.

⁹ For Australia, the monthly CPI is not available because both the Reserve Bank of Australia and the Australian Bureau of Statistics only publish quarterly data. The monthly series was obtained using linear interpolation of the CPI index.

¹⁰ The industrial production cycle had to be used as a proxy for the output gap given that GDP data are not available at monthly frequency. Though a bit more volatile, it is highly correlated with the output gap from GDP (comparison at quarterly frequency). Moreover, industrial production data tend to be revised less often and to a lesser extent than the GDP data, which reduced the problem of real-time vs. ex-post data present in the GDP data.

We proxy financial stress by means of the FSI provided recently by the IMF (Cardarelli et al., 2011), which is a consistent measure for a wide range of countries but, at the same time, is sufficiently comprehensive to track stress of a different nature. It includes the main components of financial stress in an economy and is available for a reasonably long period to be used for our empirical analysis (see Figure 1). We use both the overall index, which is a sum of seven components, as well as each sub-index and component separately:

(i) Banking-related sub-index components: the inverted term spread (the difference between short-term and long-term government bonds), TED spread (the difference between interbank rates and the yield on Treasury bills), banking beta (12-month rolling beta, which is a measure of the correlation of banking stock returns to total returns in line with the CAPM);

(ii) Securities-market-related sub-index components: corporate bond spread (the difference between corporate bonds and long-term government bond yields), stock-market returns (monthly returns multiplied by -1), time-varying stock-return volatility from the GARCH(1,1) model;

(iii) Foreign-exchange-related sub-index: the time-varying volatility of monthly changes in NEER, from the GARCH (1,1) model.

We examined various alternative methods of aggregating the components – simple sum, varianceequal weighting, and PCA weighting – but failed to uncover any systematic differences among these in terms of the values of the overall index and consecutively in the empirical results. Cardarelli et al. (2011) confirm that extreme values of this indicator correctly identify almost all (approximately 80%–90%) of the financial crises (including banking, currency, and other crises, along with stock and house-price boom and busts) identified in previous studies.

The use of a composite index has a number of benefits. First, it approximates the evolution of financial stress caused by different factors and thus is not limited to one specific type of instability. Second, the inclusion of additional variables in the stress index does not affect the evolution of the indicator markedly (Cardarelli et al., 2011). Third, the composition of the indicator allows for breaking down the reactions of the central bank with respect to different stress subcomponents. Nevertheless, one has to be cautious about the interpretation. The composite indicator might suggest a misleading interpretation as long as the stress is caused by variables not included in the FSI but rather highly correlated with some subcomponent. An example is the case of Sweden during the ERM crisis. At the time of the crisis, Sweden maintained a fixed exchange rate, and the Riksbank sharply increased interest rates to sustain the parity. However, this is not captured by the exchange-rate subcomponent of the FSI, which measures exchange-rate volatility, because the volatility was actually close to zero. A closer examination of the data shows that this period of stress is captured by the inverted term structure; hence, it is incorrectly attributed to bank stress. A similar pattern can be observed for the UK, where the FSI increases after the announcement of withdrawal from the ERM.



Figure 1: IMF Financial Stress Indicator

Note: The figure presents the evolution of the IMF stress index over time. Higher numbers indicate more stress (see Cardarelli et al., 2011).

3.2 The Empirical Model

Following Clarida et al. (1998, 2000), most empirical studies assume that the central bank sets the nominal interest rate in line with the state of the economy typically in a forward-looking manner:

$$r_{t}^{*} = \overline{r} + \beta \left(E \left[\pi_{t+i} \left| \Omega_{t} \right] - \pi_{t+i}^{*} \right) + \gamma E \left[y_{t+j} \left| \Omega_{t} \right] \right]$$

$$\tag{1}$$

where r_t^* denotes the targeted interest rate, \overline{r} is the policy neutral rate¹¹, π_{t+i} stands for the central bank forecast of the yearly inflation rate, *i* indicates periods ahead based on an information set Ω_t used for interest-rate decisions available at time *t*, and π_{t+i}^* is the central bank's inflation target.¹² y_{t+i} represents a measure of the output gap.

Nevertheless, Eq. (1) was found to be too restrictive to provide a reasonable description of actual interest-rate setting. Notably, it does not account for interest-rate smoothing by central banks, in particular the practice whereby the central bank adjusts the interest rate sluggishly to the targeted value. This is tracked in empirical studies by the simple partial-adjustment mechanism:

$$r_{t} = \rho r_{t-1} + (1 - \rho) r_{t}^{*}$$
⁽²⁾

where $\rho \in [0,1]$ is the smoothing parameter. There is an ongoing controversy as to whether this parameter represents genuine policy inertia or reflects empirical problems related to omitted variables, dynamics or shocks (see, e.g., Rudebusch, 2006). The linear policy rule in Eq. (1) can be obtained as the optimal monetary-policy rule in the LQ framework, where the central bank aims only at price stability and economic activity. Bauducco et al. (2008) propose an NK model with a financial system where the central bank has privileged information (given its supervisory function) on the health of the financial sector. In such a setting, the common policy rule represented by Eq. (1) will be augmented by variables representing the health of the financial sector. Following this contribution, we consider the forward-looking rule where central banks may respond to a comprehensive measure of financial stress rather than stress in a particular segment (Bulíř and Čihák, 2008). In practice, the augmented rule can be of some interest to outsiders because inflation expected by the individual monetary-policy committee members is unobservable to the public (even though some central banks publish figures that may be very close to the unobserved expected inflation, such as staff inflation forecasts or inflation forecasts stemming from interactions between staff and monetary-policy committee members). In such case, outsiders may benefit from including additional indicators such as financial stress in the policy rule to predict the central bank's behavior more accurately.

Therefore, we substitute Eq. (2) into Eq. (1), assuming rational expectations we replace unobserved forecast variables by their actual realization (this includes endogeneity in the model, which is dealt with by instrumental variable estimation; see the details below), pass the inflation forecast to the generic intercept α , and include the measures of financial stress described above, which results in Eq. (3):

$$r_{t} = (1 - \rho) \left[\alpha + \beta \left(\pi_{t+i} - \pi_{t+i}^{*} \right) + \gamma y_{t+j} \right] + \rho r_{t-1} + \delta x_{t+k} + \varepsilon_{t}$$
(3)

While in Eq. (1) the term α coincides with the policy-neutral rate \overline{r} , its interpretation is not straightforward once the model is augmented by additional variables. Note that the financial

¹¹ The policy-neutral rate is typically defined as the sum of the real equilibrium rate and expected inflation.

¹² An explicit definition of an inflation target exists only for countries with an inflation-targeting (IT) regime. Most empirical studies assume, in line with Taylor (1993), that this target does not vary over time and can be omitted from the empirical model.

stress index x_{t+k} does not appear within the square brackets because it is not a variable that determines the target interest rate r_t^* , but it is rather a factor such as the lagged interest rate, i.e., it may explain why the actual interest rate r_t deviates from the target. Moreover, by placing it in the regression at the same level as a lagged interest rate, we can directly test whether this variable representing ad-hoc policy decisions decreases the interest-rate inertia ρ , as suggested by Mishkin (2009). At the same time, the response on the coefficient δ can increase, as central banks are more likely to react to financial stress when stress is high. Consequently, it is possible that ρ and δ move in opposite directions because the central bank either smoothes the interest-rate changes or adjusts the rates in the face of financial stress. In the latter case, the response is likely to be quick and substantial. We set *i* equal to 6, *j* equal to 0 and *k* equal to -1.¹³ Consequently, the disturbance term ε_t is a combination of forecast errors and is thus orthogonal to all information available at time t (Ω_t).

The empirical studies on monetary-policy rules have moved from using time-invariant estimates (Clarida et al., 1998) through sub-sample analysis (Taylor, 1999; Clarida et al., 2000) toward more complex methods that allow an assessment of the evolution of the conduct of monetary policy. Indeed, regime change has been a significant feature of monetary policy conduct over recent decades. There are two alternative methods for modeling structural changes in monetarypolicy rules that occur on an unknown date: (i) regime-switching models, in particular statedependent Markov switching models (Valente, 2003; Assenmacher-Wesche, 2006; Sims and Zha, 2006) and (ii) state-space models, where the changes are characterized by smooth transitions rather than abrupt switches (Boivin, 2006; Kim and Nelson, 2006; Trecroci and Vassalli, 2009). As argued in Baxa et al. (2010), we consider the second approach to be preferable for the estimation of policy rules, given that it is more flexible and allows for the incorporation of a simple correction of endogeneity (Kim, 2006; Kim and Nelson, 2006), which is a major issue in forward-looking policy rules estimated from ex-post data.¹⁴ The state-space approach, or timevarying coefficient model, also seems suitable when one wants to evaluate the effect of factors such as financial stress that can, for a limited length of time, alter (rather than permanently change) monetary-policy conduct.

¹³ More precisely, *i* equals 6 when we use monthly data and 2 for quarterly data. Although the targeting horizon of central banks is usually somewhat longer (4–8 quarters), as in the other papers in this stream of literature, we prefer to proxy inflation expectations by inflation in t + 2 quarters for the following reasons. First, the endogeneity correction requires a strong correlation between the endogenous regressor and its instruments. Second, the prediction error logically increases at longer horizons. Most importantly, the choice of *i* is in line with the theory. Batini and Nelson (2001) show that i = 2 in their baseline model of an optimal policy horizon. However, alternative specifications of their model show some sensitivity in terms of what is the optimal *i*. In the case of the output gap, we instead assume a backward-looking reaction. The reason is that in the absence of real-time data, we have to rely on the output-gap construction of statistical methods. It is arguable that aside from the prediction error, there is also a construction error that might be magnified if an unobserved forecast is substituted by the output-gap estimate for future periods. Finally, we assume that central bankers' response (if any) to financial stress is rather immediate (see Mishkin, 2009). Therefore, we use one lag of the FSI and its subcomponents in the benchmark case. However, as a robustness check, we allow for different lags and leads, allowing the central bankers' response to be preemptive rather than reactive.

¹⁴ The time-varying parameter model with specific treatment of endogeneity is still relevant when real-time data are used (Orphanides, 2001). The real-time forecast is not derived under the assumption that nominal interest rates will remain constant within the forecasting horizon (Boivin, 2006) or in the case of measurement error and heteroscedasticity (Kim et al., 2006).

State-space models are commonly estimated by means of a maximum likelihood estimator via the Kalman filter or smoother. Unfortunately, this approach has several limitations that can become problematic in applied work. First, the results are somewhat sensitive to the initial values of the parameters, which are usually unknown, especially in the case of variables whose impacts on the dependent variable are not permanent and whose sizes are unknown, which is the case for financial stress and its effect on interest rates. Second, the log likelihood function is highly non-linear, and in some cases optimization algorithms fail to minimize the negative of the log likelihood. In particular, it can either fail to calculate the Hessian matrix throughout the iteration process, or, when the likelihood function is approximated to facilitate computations, the covariance matrix of observation vectors can become singular for the starting values provided. The alternative is a moment-based estimator proposed by Schlicht (1981, 2005) and Schlicht and Ludsteck (2006), which is employed in our paper and briefly described below. This framework is sufficiently flexible such that it incorporates the endogeneity correction proposed by Kim (2006).

Kim (2006) shows that the conventional time-varying parameter model delivers inconsistent estimates when explanatory variables are correlated with the disturbance term and proposes an estimator of the time-varying coefficient model with endogenous regressors. Endogeneity may arise not only in forward-looking policy rules based on ex-post data (Kim and Nelson, 2006; Baxa et al., 2010) but also in the case of variables that have a two-sided relationship with monetary policy. Financial stress unquestionably enters this category. Following Kim (2006), we rewrite Eq. 3 as follows:

$$r_{t} = (1 - \rho_{t}) \left[\alpha_{t} + \beta_{t} \left(\pi_{t+i} \right) + \gamma_{t} y_{t+j} \right] + \rho_{t} r_{t-1} + \delta_{t} x_{t+k} + \varepsilon_{t}$$

$$\tag{4}$$

$$\alpha_{t} = \alpha_{t-1} + \mathcal{G}_{1,t}, \ \mathcal{G}_{1,t} \sim i.i.d.N(0,\sigma_{\mathcal{G}_{1}}^{2})$$
(5)

$$\boldsymbol{\beta}_{t} = \boldsymbol{\beta}_{t-1} + \boldsymbol{\vartheta}_{2,t}, \ \boldsymbol{\vartheta}_{2,t} \sim i.i.d.N\left(\boldsymbol{0}, \boldsymbol{\sigma}_{\boldsymbol{\vartheta}_{2}}^{2}\right)$$
(6)

$$\gamma_t = \gamma_{t-1} + \mathcal{G}_{3,t}, \ \mathcal{G}_{3,t} \sim i.i.d.N\left(0,\sigma_{\mathcal{G}_3}^2\right)$$
(7)

$$\delta_t = \delta_{t-1} + \mathcal{G}_{4,t}, \ \mathcal{G}_{4,t} \sim i.i.d.N(0,\sigma_{\mathcal{G}_4}^2)$$
(8)

$$\rho_t = \rho_{t-1} + \vartheta_{5,t}, \ \vartheta_{5,t} \sim i.i.d.N\left(0, \sigma_{\vartheta_5}^2\right) \tag{9}$$

$$\pi_{t+i} = Z'_{t-m} \xi + \sigma_{\varphi} \varphi_t, \ \varphi_t \sim i.i.d.N(0,1)$$
⁽¹⁰⁾

$$y_{t+j} = Z'_{t-m} \psi + \sigma_v v_t, \ v_t \sim i.i.d.N(0,1)$$
(11)

$$x_{t+k} = Z_{t-m} o + \sigma_t t_t, \ t_t \sim i.i.d.N(0,1)$$
(12)

The measurement Eq. (4) of the state-space representation is the monetary-policy rule. The transitions in Eqs. (5)–(9) describe the time-varying coefficients as a random-walk process without drift.¹⁵ Eqs. (10)–(12) track the relationship between the potentially endogenous regressors (π_{t+i} , y_{t+i} , and x_{t+k}) and their instruments, Z_t . We use the following instruments:

¹⁵ Note that while a typical time-invariant regression assumes that $a_t = a_{t-1}$, in this case, it is assumed that $E[a_t] = a_{t-1}$.

 π_{t-1} , π_{t-12} (π_{t-4} for CAN and SWE), y_{t-1} , y_{t-2} , r_{t-1} , the foreign interest rate for countries other than the United States (the three-month EURIBOR for SWE and UK, and the US three-month interbank rate for CAN and AUS). Unlike Kim (2006), we assume that the parameters in Eqs. (10)–(12) are time-invariant. The correlation between the standardized residuals φ_t , v_t , and t_t and the error term ε_t is $\kappa_{\varphi,\varepsilon}$, $\kappa_{v,\varepsilon}$, and $\kappa_{t,\varepsilon}$, respectively (note that σ_{φ} , σ_v , and σ_t are the standard errors of φ_t , v_t , and t_t , respectively). Consistent estimates of the coefficients in Eq. (4) are obtained in two steps. In the first step, we estimate Eqs. (10)–(12) and save the standardized residuals φ_t , v_t , and t_t . In the second step, we estimate Eqs. (10)–(12) and save the standardized residuals φ_t , v_t , and t_t . In the second step, we estimate Eqs. (13) below along with Eqs. (5)–(9). Note that Eq. (13) now includes bias correction terms, i.e., the (standardized) residuals from Eqs. (10)–(12), to address the aforementioned endogeneity of the regressors. Consequently, the estimated parameters in Eq. (13) are consistent, as ζ_t is uncorrelated with the regressors.

$$r_{t} = (1 - \rho_{t}) [\alpha_{t} + \beta_{t} \pi_{t+6} + \gamma_{t} \gamma_{t-1}] + \rho_{t} r_{t-1} + \delta_{t} x_{t-1} + \kappa_{\varphi,\varepsilon} \sigma_{\varepsilon} \varphi_{t} + \kappa_{\nu,\varepsilon} \sigma_{\varepsilon} \nu_{t} + \kappa_{\iota,\varepsilon} \sigma_{\varepsilon} \iota_{t} + \zeta_{t} \zeta_{t} - N (0, (1 - \kappa_{\varphi,\varepsilon}^{2} - \kappa_{\nu,\varepsilon}^{2} - \kappa_{\iota,\varepsilon}^{2}) \sigma_{\varepsilon,t}^{2})$$

$$(13)$$

As previously noted, instead of the standard framework for second-step estimation, the maximum likelihood estimator via the Kalman filter (Kim, 2006), we use an alternative estimation framework, the "varying coefficients" (VC) method (Schlicht, 1981; Schlicht, 2005; Schlicht and Ludsteck, 2006). This method is a generalization of the ordinary least squares approach that, instead of minimizing the sum of the squares of the residuals $\sum_{t=1}^{T} \zeta^2$, uses minimization of the weighted sum of the squares:

$$\sum_{t=1}^{T} \zeta^2 + \theta_1 \sum_{t=1}^{T} \vartheta_1^2 + \theta_2 \sum_{t=1}^{T} \vartheta_2^2 + \ldots + \theta_n \sum_{t=1}^{T} \vartheta_n^2$$
(14)

where the weights θ_i are the inverse variance ratios of the regression residuals ε_t and the shocks in time-varying coefficients ϑ_t , that is, $\theta_i = \sigma^2 / \sigma_i^2$. This approach balances the fit of the model and parameter stability. Additionally, the time averages of the regression coefficients, estimated by a weighted least squares estimator, are identical to their GLS estimates of the corresponding regression with fixed coefficients, that is, $\frac{1}{T} \sum_{t=1}^{T} \hat{a}_t = \hat{a}_{GLS}$.¹⁶ The method is useful in our case because:

- it does not require knowledge of initial values even for non-stationary variables prior to the estimation procedure. Instead, both the variance ratios and the coefficients are estimated simultaneously;
- the property of the estimator that the time averages of the estimated time-varying coefficients are equal to its time-invariant counterparts, permits easy interpretation of the results in relation to time-invariant results;

¹⁶ See Schlicht and Ludsteck (2006) and Baxa et al. (2010) for more details.

• it coincides with the MLE estimator via the Kalman filter if the time series are sufficiently long and if the variance ratios are properly estimated.¹⁷ However, this method suffers from certain limitations of its own. In particular it requires that: (a) the time-varying coefficients are described as random walks, and (b) the shocks in time-varying coefficients \mathcal{P}_t are minimized (see Eq. (14)).

While this does not represent a major problem for the estimation of the coefficients of common variables such as inflation, where the monetary-policy response is permanent, it can lead to a loss of some information about ad-hoc response factors in monetary policy making that are considered by central bankers only infrequently; however, once they are in place, the policy response can be substantial. The financial stress indicator x_{t+k} seems to be this kind of factor. One way to address this problem is by estimation-independent calibration of the variance ratios in Eq. (14), such that the estimated coefficient is consistent with economic logic, i.e., it is mostly insignificant and can become significant (with no prior restriction on its sign) during periods of financial stress, i.e., when the financial stress indicator is different from zero. Therefore, we first estimate Eq. (13) using the VC method and study whether the resulting coefficients in the FSI correspond to economic intuition, especially whether the coefficient is not constant or slowly moving (the socalled pile-up problem, see Stock and Watson, 1998). When this problem occurs, we compare the results with models where k belongs to (-2, -1, 0, 1, 2) and calibrate the variance ratios in Eq. (13) by the variance ratios estimated for the model with the largest variances in the FSI. This step was necessary for Australia and Sweden. The Taylor-rule coefficients were compared with the initial estimates and were consistent in both cases.¹⁸

The results of our empirical analysis should reveal whether central banks adjusted their interestrate policies in the face of financial stress. However, the time-varying framework also allows for inferring whether any response to financial stress led to the temporal dismissal of other targets, in particular the inflation rate. Therefore, we are mainly interested in the evolution of the financialstress coefficient δ_t . We expect it to be mostly insignificant or zero, given that episodes of financial stress are rather infrequent, and even if they occur, the monetary authorities may not always respond to them. Moreover, the size of the estimated coefficient does not have any obvious interpretation because the FSI is a composite indicator normalized to have a zero mean. Consequently, we define the stress effect as a product of the estimated coefficient δ_t and the value of the IMF's FSI x_{t+k} . The interpretation of the stress effect is straightforward: it shows the magnitude of interest-rate reactions to financial stress in percentage points or, in other words, the

¹⁷ The Kalman filter as implemented in common econometric packages typically uses the diffusion of priors for its initiation, but it still produces many corner solutions and often does not achieve convergence. Schlicht and Ludsteck (2006) compare the performance of the moment estimator and the Kalman smoother in terms of the mean squared error on simulated data, and they conclude that the moment estimator outperforms the Kalman filter on small samples with a size of up to 100 observations. For comparison, we estimated Eq. (12) using the conventional Kalman filter in the GROCER software using the tvp function (Dubois-Michaux, 2009). We parameterized the model by initial conditions taken from the OLS estimates of the parameters on the full sample and the initial forecast error covariance matrix set to 0. The matrix of the residuals of time-varying coefficients is assumed to be diagonal, as in the VC method. The results were very similar to those obtained from the VC method when the estimated variances were the same in both methods.

¹⁸ Stock and Watson (1998) propose a medium-unbiased estimator for variance in the time-varying parameter model, but its application is straightforward only in the case of one time-varying coefficient, and more importantly, it requires the variables to be stationary.

deviation from the target interest rate, as implied by the macroeconomic variables, due to the response to financial stress.

4. Results

This section summarizes our results on the effect of financial stress on interest-rate setting. First, the results on the effect of the overall measure of financial stress on interest-rate setting are presented. Second, the effect of specific components of financial stress on monetary policy is examined. Third, we briefly comment on the monetary-policy rule estimates that served as the input for the assessment of financial-stress effects. Finally, we perform a series of robustness checks.

4.1 Financial-Stress Effect

Figure 2 presents our results on the effect of financial stress on interest-rate setting in all five countries (referred to as the *financial-stress effect* hereinafter).¹⁹ Although there is some heterogeneity across countries, some global trends in the effect of financial stress are apparent. Whereas in good times, such as in the second half of the 1990s, financial stress has virtually no effect on interest-rate setting or is slightly positive,²⁰ the reaction of monetary authorities to financial stress was highly negative during the 2008–2009 global financial crisis. While the previous evidence on the effect of financial stress on monetary policy is somewhat limited, our results broadly confirm the time-invariant findings of Cecchetti and Li (2008), who show that the US Fed adjusted interest rates to the procyclical impact of bank capital requirements in 1989–2000. Similarly, Belke and Klose (2010) estimate the Taylor rule on two sub-samples (before and during the 2008–2009 global financial crisis) and find that the Fed reacted systematically not only to inflation and the output gap, but also to asset prices, credit, and money.

¹⁹ Given that the magnitude of the financial-stress effect differs across countries, especially due to the high positive peak for Sweden and negative peak for the UK, we use different scales for different countries.

²⁰ Note that the positive effect of financial stress on interest-rate setting is to some extent a consequence of scaling the financial-stress indicator; its zero value corresponds to the long-run average stress. Hence, we do not pay much attention to positive values of stress unless caused by a temporarily positive and significant regression coefficient associated with the FSI.



Figure 2: The Effect of Financial Stress on Interest-Rate Setting

Notes: The figure depicts the evolution of the financial-stress effect. The stress effect (y-axis) is defined as the product of the estimated coefficient on the financial-stress indicator in the monetary-policy rule and the value of the IMF financial-stress indicator (δx). The stress effect shows the magnitude of the interest-rate reaction to financial stress in percentage points.

The size of financial-stress effects on interest-rate setting during the recent financial crisis is somewhat heterogeneous, with the strongest reaction found for the UK. The results suggest that all central banks except the Bank of England maintain policy rates at approximately 50–100 basis points lower compared to the counterfactual policy of no reaction to financial stress. The size of

this effect for the UK is assessed to be approximately three times stronger (i.e., 250 basis points). This implies that approximately 50% of the overall policy-rate decrease during the recent financial crisis was motivated by financial-stability concerns in the UK (10%–30% in the remaining sample countries), while the remaining half falls to unfavorable developments in domestic economic activity. This finding complements previous results suggesting that the BoE's consideration of expected inflation over the last decade has been very low (as found by Baxa et al., 2010, using the time-varying model and by Taylor and Davradakis, 2006, in the context of the threshold model) by evidence that it further decreased during the current crisis. It is also evident that the magnitude of the response is unusual for all five central banks. However, the results for Australia, Canada, and Sweden show a similar magnitude of response to financial stress during the recent financial crisis compared to that observed in previous periods of high financial stress.

Given that the 2008–2009 global crisis occurred at the end of our sample (there is a peak in the stress indicator of five standard deviations that has not returned to normal values yet), we performed an additional check to avoid possible end-point bias. In particular, we ran our estimation excluding the observation from the period of the 2008–2009 crisis. These results were practically indistinguishable from the full sample estimation. With regard to the effect of the current crisis, the largest uncertainty is associated with the results for Canada, for which the shortest data sample – ending in the fourth quarter of 2008 – was available. When the possibility of a preemptive reaction of the central bank to financial stress is considered (see the robustness checks below), the effect of financial stress in the current crisis is estimated for Canada at somewhere between 1% and 2% (see Appendix 3). These additional results suggest that the response of the Bank of Canada in the benchmark model is likely to be underestimated.

The question of which components of financial stress influence interest-rate setting is addressed in Figure 3. In this case, we estimate the model using each FSI subcomponent separately (the bank stress effect, the exchange-rate stress effect, and the stock-market stress effect) instead of the overall FSI and report the financial-stress effect attributable to each subcomponent. Some heterogeneity across countries is again apparent, although it seems that bank stress and stockmarket stress dominated central bankers' considerations in less open economies. On the other hand, exchange-rate stress matters in more open economies such as Canada and Sweden.

Specifically, the US Fed seemed to be worried about financial instability, especially during the 1980s. We can observe that the main concern in the early 1980s was banking stress, which is arguably related to the Savings and Loans crisis. Another concern was that of stock-market stress, in particular during the stock-market crash of 1987, when interest rates were 30 b.p. lower with respect to the benchmark case.

The Bank of England was, in general, much more perceptive to financial stress. We find its response mainly to stock-market stress again, notably, in 1987. Interestingly, we find little response to exchange-rate stress, not even during the 1992 ERM crisis. Nevertheless, it has to be emphasized that the interest-rate reaction to this speculative attack was subdued in comparison to, for example, the Riksbank (Buiter et al., 1998). The base rate was increased by 2 p.p. to 12% on September 16, 1992. Despite a promise of further increases up to 15%, traders continued selling the pound. On the evening of the same day, the UK left the ERM with interest rates unchanged;

on the following day, the base rate decreased to 10.5%; and at the end of September, the base rate was 9%, lower than at the beginning of the month. Therefore, despite huge open market operations, the response of the interest rate was moderate, with the monthly interest-rate average practically unaffected. Hence, our framework does not detect any effect of financial stress on the interest rate during the ERM crisis. Since the devaluation of the pound sterling in September 1992, the effect of financial stress on interest-rate setting approaches zero from originally negative values. Aside from this, the response of the Bank of England to inflation has decreased. From this perspective, it seems the pound sterling's withdrawal from the ERM allowed for both a more rule-based and less restrictive monetary policy. With respect to the banking crisis in the late 2000s, the Bank of England provided liquidity support in its earlier stage in 2007 with the fall of Northern Rock. Policy rates remained constant until late 2008, despite the bankruptcy of Lehman Brothers in the US in September 2008. The reason for keeping policy rates constant was related to concerns regarding potential inflationary pressures from rising oil and food prices.

The interest-rate effect of the banking crisis in Sweden in the early 1990s is estimated to be slightly over 1% in absolute terms (see Figure 2). The crisis began in September 1990, when the non-banking financial institution Nyckeln unexpectedly collapsed (Jennergren, 2002). The Riksbank did not decrease interest rates sharply because coincidental international factors, in particular the reunification of Germany, forced interest rates upwards. Despite facing recession, the government attempted to defend the peg of the krona to ECU and decided to prevent the spread of the banking crisis by announcing a blanket guarantee for the liabilities of the banking sector (Jonung, 2009). Hence, interest-rate cuts were not a primary tool chosen for resolution of the crisis.

In comparison to the United Kingdom, the reaction of the Riksbank to the ERM crisis was different. First, after a series of speculative attacks on the Swedish krona in mid-September 1992, the Riksbank still attempted to maintain the fixed exchange rate, and the marginal interest rate jumped up 500% to offset the outflow of liquidity and other speculative attacks (see the large positive stress effect on the interest rate in 1992 in Figure 2). However, not even such an increase was sufficient, and the fixed exchange rate had to be abandoned later, in November.²¹

The Reserve Bank of Australia significantly loosened its policy during the 1980s. This can be attributed to stress in the banking sector with the exception of the reaction to the stock-market crash in 1987 (see Figure 3).

The exchange rate as well as bank stress seems to matter for interest-rate considerations at the Bank of Canada. Interestingly, the results suggest that the Bank of Canada often responded to higher exchange-rate stress by monetary tightening. A possible explanation for this finding might be that given the openness of the Canadian economy, its central bank tightened the policy when the currency stabilized at the level that the monetary authority considered to be undervalued.

²¹ For Sweden, we add a dummy variable for the third quarter of 1992 (ERM crisis) to Eq. 13. At this time, the Swedish central bank forced short-term interest rates upward in an effort to keep the krona within the ERM. From the perspective of our model, it was a case of a strong positive reaction to the actual stress that lasted only one period. When this dummy variable was not included, the model with a lagged value of the FSI was unable to show any link between stress and interest rates, and the estimates of other coefficients were inconsistent with economic intuition.

We would like to highlight a comparison of Figures 2 and 3. First, it should be noted that a positive response to one stress subcomponent may cancel out in the face of a negative response to another one, making the response to the overall stress negligible (as in the case of Canada). Second, the stress effects related to individual subcomponents do not necessarily sum up to the stress effect related to the entire FSI.

Overall, the results suggest that the central bank tends to react to financial stress, and different components of financial stress matter in different time periods. The effect of financial stress on interest-rate setting is found to be virtually zero in good times and economically sizable during periods of high financial stress.

Figure 3: The Effect of Financial Stress Components on Interest-Rate Setting: Bank Stress, Exchange-Rate Stress, and Stock-Market Stress



Notes: The figure depicts the evolution of the components of the financial-stress effect, namely, the bankstress effect, the exchange-rate stress effect, and the stock-market stress effect. The stress effect (yaxis) is defined as the product of the estimated coefficient on the given component of the financialstress indicator in the monetary-policy rule and the value of the corresponding component of the IMF financial-stress indicator (δx). The stress effect shows the magnitude of the interest-rate reaction to financial stress in percentage points.

4.2 Monetary Policy Rule Estimates

Given that our main interest lies in the interest-rate response to financial stress, we comment on the other monetary-policy rule estimates only briefly. The plot of the evolution of the estimated parameters over time for all countries is available in Appendix 1. First of all, it should be noted that most coefficients do indeed vary over time, which is consistent with previous evidence and underlines the fact that monetary-policy conduct has evolved substantially in recent decades.

In general, the responses to inflation (β) are positive, and the coefficient is often above one, consistent with the Taylor principle. Nevertheless, we find that in the last decade the coefficient decreased somewhat, and during the recent financial crisis it even turned slightly negative (in the US and UK; more on this below). The decrease of the inflation response during the last decade is typically attributed to well-anchored inflation expectations as well as a low-inflation environment (Sekine and Teranishi, 2008; Baxa et al., 2010). The finding of negative β during the recent crisis is likely to be related to the fact that central banks were decreasing policy rates to historical lows in the face of exceptionally high financial stress, despite inflation expectations being largely unchanged, rather than being an indication that policy rates were systematically decreased when inflation expectations increased.²²

For the United States, our results show that the response to inflation was highest in the early 1980s, and except for the period following the recession of 1990–1991 the estimated coefficient is higher or very close to one. This value is slightly lower in comparison to Kim and Nelson (2006), who found the response to be around 1.5 and almost invariant since 1981. Given the size of the confidence intervals, it is, however, difficult to determine whether our results differ significantly. Kim and Nelson (2006) estimate the interest-rate smoothing coefficient to be higher than 0.8, i.e., in line with what time-invariant estimates of monetary-policy rules typically suggest (see, for example, Clarida et al., 1998, 2000). Our estimates indicate that the interest-rate smoothing is somewhat lower (0.5-0.6). This finding is in line with the recent critique by Rudebusch (2006), who argues that the practical unpredictability of interest-rate changes over a few quarters suggests that the degree of interest-rate smoothing is rather low. Interestingly, we find that the response to inflation decreases substantially after the terrorist attacks on September 11, 2001. This complies with Greenspan (2007), who argued in that case that the Fed was concerned about the US economy spiraling downward into recession after the terrorist attacks. Later, Greenspan himself acknowledged that the monetary policy was somewhat loose, but ex ante optimal, given the increased uncertainty after the attacks. In a similar vein, Taylor (2010) compares the actual values of the federal funds rate and the counterfactual values predicted by the (time-invariant) Taylor rule, finding that in 2002–2005 interest rates were too low compared to predictions and this deviation from a rules-based policy was "larger than in any period since the unstable decade before the Great Moderation" (p. 167). Negative estimates of the response to inflation in this particular period are reported also by Trecroci and Vassalli (2010). The response to the output gap is significant for nearly the whole sample, although the values

²² It should also be kept in mind that strictly speaking some central banks in the determined period did not "follow a rule" due to institutional constraints (e.g. the BoE's dependence on the government in the 1980s). Therefore, our time-varying estimates can be also interpreted as covariates of interest rate dynamics.

close to 0.2 are somewhat lower than in Kim-Nelson (2007), but similar to Trecroci and Vassalli (2010).

The results for countries that currently have an explicit target for inflation share several features. The interest-rate smoothing is again found to be lower in comparison to time-invariant estimates, with midpoints around 0.5. The exception is Canada, where the values fluctuate around zero and are insignificant. Moreover, for some central banks, such as the RBA and the BoE in 2010 or the Sveriges Riksbank in late 1980, we find support for Mishkin's (2009, 2010) argument that central banks are less inertial during crises.²³ Second, the response of interest rates to inflation is particularly strong during the periods when central bankers want to break a record of high inflation, such as in the UK or Australia at the beginning of the 1980s, and is less aggressive in a low-inflation environment with subdued shocks and well-anchored inflation expectations (Kuttner and Posen, 1999). In this respect, our results confirm the findings of Taylor and Davradakis (2006), who argue that the response of the Bank of England to inflation is insignificant when the inflation rate is close to its target. Third, some central banks (Australia and Canada) are also found to react to output-gap developments, with the parameter estimated to be slightly positive on average, whereas the parameter is insignificant with wide confidence intervals in Sweden and the United Kingdom. It is arguable whether consideration of real economic activity should be interpreted as a practice of more flexible inflation targeting or is only an indication that the output gap predicts inflation pressures on the supply side.

The results show that the interest-rate response to financial stress is insignificant most of the time, at the 95% significance level. This is in line with our expectations, i.e., that the coefficients should be insignificant in periods when stress is low. Nevertheless, the coefficient on financial stress is statistically significant at the 95% level during the recent financial crises for most countries. The importance of financial stress for interest-rate setting is further confirmed using the GMM estimation, which shows that the financial-stress index is significant, in fact, in all countries. In addition, when the one-standard-deviation quantile is taken into account instead of the more usual two-standard-deviation quantile, the periods when we can identify any interest-rate response to financial stress become more evident. We present a list of these periods in Table A5.2 in the Appendix.

4.3 Robustness Checks

In terms of the financial-stress effect estimates, we perform a battery of robustness checks. First, following the argument put forward above that the interbank rate may occasionally provide a better signal of monetary-policy intentions than the policy rate, we use interbank interest rates as a dependent variable. These results are reported in Figures A.2.1–A.2.2. We can observe that the overall stress effect on the interbank rate was larger for the US during the current crisis, where it explains 2% of the decrease of the interbank interest rate. For Sweden, we found a strong positive effect of exchange rate volatility in the late 1980s; this might be linked to the aim of the central

 $^{^{23}}$ Indeed, the correlation coefficient of the estimated time-varying coefficient of the lagged interest rate ρ and the financial-stress index δ is -0.79 for Australia, 0.21 for Canada, -0.20 for Sweden, -0.68 for the UK, and 0.60 for the US.

bank to keep the exchange rate fixed. In other cases, there is no substantial difference between the benchmark results and the results obtained using this alternative dependent variable.

Second, in the benchmark model and all of the results reported thus far, we use the first lag of the FSI in the policy-rule estimation. We motivate this choice by the use of monthly data, the frequency of monetary-policy meetings of most central-bank boards, and the assumption that policy actions are likely to be implemented in a timely fashion. In addition, we employ different lags and leads, in the latter case allowing the policy to be preemptive rather than reactive. In this case, we use the future realized value of the FSI as a proxy for the central bank's expectation (in a similar manner as to how it is routinely executed for inflation expectations) and, consequently, treat the FSI as an endogenous variable (see Figure A.3.1 for the results). To obtain comparable results, we calibrate the variance ratios with the same values as in the baseline specification. Although we find rather mixed evidence on preemptive policy actions, which may also be related to the inadequacy of proxying the expected values of financial stress by the actual values of the stress preemptively, the reaction to financial stress in the current crisis is strongly negative for both expected and observed stress.

Third, we further break down the FSI sub-indices to each underlying variable to evaluate their individual contributions.²⁴ The corresponding stress effects appear in Figures A.4.1–A.4.2. Breaking down stock-market-related stress, we find that the US Fed and the BoC react to the corporate bond spread, whereas the BoE and Sveriges Riksbank are more concerned with stock returns and volatility. While the RBA seems to be concerned with both corporate bond spreads and stock-market volatility in the 1980s, the role of stock-related stress had substantially decreased by then. As far as bank-related stress is concerned, the TED spread plays a major role in all countries apart from the UK, where the largest proportion of the effect on the interest rate can be attributed to an inverted term structure.

Fourth, because the verifications related to comparing our econometric framework to obvious alternatives such as, first, the use of a maximum likelihood estimator via the Kalman filter instead of the moment-based time-varying coefficient framework of Schlicht and, second, the use of a Markov switching model instead of a state-space model, were provided in Baxa et al. (2010), we estimate simple time-invariant monetary-policy rules for each country by the generalized method of moments, including various subsamples. This simple evidence reaffirms that the analyzed central banks seem to pay attention to overall financial stress in the economy. The FSI is statistically significant, with a negative sign and a magnitude of between 0.05–0.20 for all countries. On the other hand, the coefficients of its subcomponents often are not significant, and the exchange-rate subcomponent in some cases has a positive sign. These results, which are available upon request, confirm that to understand the interest-rate adjustment in response to financial stress, one should rely on a model allowing for a differential response across time.

²⁴ This applies only to the banking and stock-market subcomponents because the foreign-exchange subcomponent is represented by a single variable.

5. Concluding Remarks

The 2008–2009 global financial crisis generated significant interest in exploring the interactions between monetary policy and financial stability. This paper aimed to examine in a systematic manner whether and how the monetary policy of selected main central banks (the US Fed, the Bank of England, the Reserve Bank of Australia, the Bank of Canada, and Sveriges Riksbank) responded to episodes of financial stress over the last three decades. Instead of using individual alternative measures of financial stress in different markets, we employed the comprehensive indicator of financial stress recently developed by the International Monetary Fund, which tracks overall financial stress as well as its main subcomponents, in particular banking stress, stock-market stress and exchange-rate stress.

Unlike a few existing empirical contributions that aim to evaluate the impact of financial-stability concerns on monetary policy making, we adopt a more flexible methodology that not only allows for the response to financial stress (and other macroeconomic variables) to change over time, but also addresses potential endogeneity (Kim and Nelson, 2006). The main advantage of this framework is that it not only enables testing of whether central banks responded to financial stress at all, but also detects the periods and types of stress that were the most worrying for monetary authorities. Our results indicate that central banks truly change their policy stances in the face of financial stress, but the magnitude of such responses varies substantially over time. As expected, the impact of financial stress on interest-rate setting is essentially zero most of the time, when the levels of stress are very moderate. However, most central banks loosen monetary policy when the economy faces high financial stress. There is some cross-country and time heterogeneity when we examine central banks' considerations of specific types of financial stress. While most central banks seem to respond to stock-market stress and bank stress, exchange-rate stress is found to drive the reaction of central banks only in more open economic

Consistent with our expectations, the results indicate that a sizeable fraction of the monetarypolicy easing during the 2008–2009 financial crisis can be explained by a direct response to the financial stress above what might be attributed to the decline in inflation expectations and output below its potential. However, the size of the financial-stress effect differs by country. The result suggests that all central banks except the Bank of England kept their policy rates at 50–100 basis points lower, on average, solely due to the financial stress present in the economy. Interestingly, the size of this effect for the UK is assessed at about three times stronger (i.e., 250 basis points). This implies that about 50% of the overall policy-rate decrease during the recent financial crisis was motivated by financial-stability concerns in the UK (10%–30% in the remaining sample countries), while the remaining half falls to unfavorable developments in domestic economic activity. For the US Fed, macroeconomic developments themselves (a low-inflation environment and output substantially below its potential) explain the majority of the interest-rate policy decreases during the crisis, leaving any further response to financial stress to be constrained by zero interest rates.

Overall, our results point to the usefulness of augmenting the standard version of monetary-policy rules by some measure of financial conditions to obtain a better understanding of the interest-rate-setting process, especially when financial markets are unstable. The empirical results suggest that the central banks considered in this study altered the course of their monetary policy in the face of

financial stress. The recent crisis seems truly to be an exceptional period, in the sense that the response to financial instability was substantial and coincided in all the countries analyzed, which is evidently related to intentional policy coordination absent in previous decades. However, we have also observed that previous idiosyncratic episodes of financial distress were, at least in some countries, followed by monetary-policy responses of similar, if not higher, magnitude.

References

- ABIAD, A., R. BALAKRISHNAN, P. K. BROOKS, D. LEIGH, AND I. TYTELL (2009): "What's the Damage? Medium-term Output Dynamics after Banking Crises." IMF Working Paper No. 09/245.
- AISEN, A. AND M. FRANKEN (2010): "Bank Credit during the 2008 Financial Crisis: A Cross-Country Comparison." IMF Working Paper No. 10/47.
- AKRAM, Q. F., G. BÅRDSEN, AND K.-G. LINDQUIST (2007): "Pursuing Financial Stability under an Inflation-targeting Regime." *Annals of Finance* 3, pp. 131–153.
- AKRAM, Q. F. AND Ø. EITRHEIM (2008): "Flexible Inflation Targeting and Financial Stability: Is It Enough to Stabilize Inflation and Output?" *Journal of Banking & Finance* 32, pp. 1242–1254.
- ASSENMACHER-WESCHE, K. (2006): "Estimating Central Banks' Preferences from a Time-varying Empirical Reaction Function." *European Economic Review* 50, pp. 1951–1974.
- BALAKRISHNAN, R., S. DANNINGER, S. ELEKDAG, AND I. TYTELL (2009): "The Transmission of Financial Stress from Advanced to Emerging Economies." IMF Working Paper No. 09/133.
- BAXA, J., R. HORVÁTH, AND B. VAŠÍČEK (2010): "How Does Monetary Policy Change? Evidence on Inflation Targeting Countries." Czech National Bank Working Paper No. 2-2010.
- BATINI, N. AND E. NELSON (2001): "Optimal Horizons for Inflation Targeting." Journal of Economic Dynamics and Control 25, pp. 891–910.
- BAUDUCCO, S., A. BULÍŘ, AND M. ČIHÁK (2008): "Taylor Rule Under Financial Instability." IMF Working Paper No. 08/18.
- BELKE, A. AND J. KLOSE, 2010: "(How) Do the ECB and the Fed React to Financial Market Uncertainty? The Taylor Rule in Times of Crisis." DIW Berlin Discussion Paper No. 972.
- BERNANKE, B., M. GERTLER, AND S. GILCHRIST (1996): "The Financial Accelerator and the Flight to Quality." *The Review of Economics and Statistics* 78(1), pp. 1–15.
- BERNANKE, B., M. GERTLER, AND S. GILCHRIST (1999): "The Financial Accelerator in a Quantitative Business Cycle Framework." In: J. B. Taylor and M. Woodford (eds): Handbook of Macroeconomics. North-Holland: Amsterdam.
- BERNANKE, B. AND A. BLINDER (1992): "The Federal Funds Rate and the Channels of Monetary Transmission." *American Economic Review* 82(4), pp. 901–921.
- BERNANKE, B. AND M. GERTLER (1999): "Monetary Policy and Asset Price Volatility." *Economic Review* (FRB of Kansas City), pp. 17–51.
- BERNANKE, B. AND M. GERTLER (2001): "Should Central Banks Respond to Movements in Asset Prices?" *American Economic Review* 91(2), pp. 253–257.
- BOIVIN, J. (2006): "Has U.S. Monetary Policy Changed? Evidence from Drifting Coefficients and Real-Time Data." *Journal of Money, Credit and Banking* 38(5), pp. 1149–1173.

- BORIO, C. AND P. LOWE (2002): "Asset Prices, Financial and Monetary Stability: Exploring the Nexus." BIS Working Paper No. 114.
- BORIO, C. AND P. LOWE (2004): "Securing Sustainable Price Stability: Should Credit Come Back from the Wilderness?" BIS Working Paper No. 157.
- BORIO, C. AND I. SHIM (2007): "What Can (Macro-)Prudential Policy Do to Support Monetary Policy?" BIS Working Paper No. 242.
- BORIO, C. AND P. DISYATAT (2009): "Unconventional Monetary Policies: An Appraisal." BIS Working Paper No. 292.
- BORIO, C. AND M. DREHMANN (2009): "Towards an Operational Framework for Financial Stability: 'Fuzzy' Measurement and Its Consequences." BIS Working Paper No. 284.
- BROUSSEAU, V. AND C. DETKEN (2001): "Monetary Policy and Fears of Financial Instability." ECB Working Paper No. 89.
- BUITER, W. H., G. M. CORSETTI, AND P. A. PESENTI (1998): "Interpreting the ERM Crisis: Country-Specific and Systemic Issues." Princeton Studies in International Economics, No. 84, Princeton University.
- BULÍŘ, A. AND M. ČIHÁK (2008): "Central Bankers' Dilemma When Banks Are Vulnerable: To Tighten or Not To Tighten?" IMF mimeo.
- CARDARELLI, R., S. ELEKDAG, AND S. LALL (2011): "Financial Stress and Economic Contractions." *Journal of Financial Stability* 7(2), pp. 78–97.
- CARLSON, M. A., T. B. KING, AND K. F. LEWIS (2009): "Distress in the Financial Sector and Economic Activity." Finance and Economics Discussion Series 2009-01, Board of Governors of the Federal Reserve System.
- CECCHETTI, S., H. GENBERG, J. LIPSKY, AND S. WADHWANI (2000): "Asset Prices and Central Bank Policy." Geneva Reports on the World Economy, No. 2.
- CHADHA, J.S., L. SARNO, AND G. VALENTE (2004): "Monetary Policy Rules, Asset Prices and Exchange Rates." IMF Staff Papers, 51(3).
- CHRISTIANO, L., C. ILUT, R. MOTTO, AND M. ROSTAGNO (2008): "Monetary Policy and Stock Market Boom-Bust Cycles." ECB Working Paper No. 955.
- CHOI, W. AND D. COOK (2004): "Liability Dollarization and the Bank Balance Sheet Channel." *Journal of International Economics* 64(2), pp. 247–275.
- CHRISTIANO, L., C. ILUT, R. MOTTO, AND M. ROSTAGNO (2008): "Monetary Policy and Stock Market Boom-Bust Cycles." ECB Working Paper No. 955.
- CLARIDA, R., J. GALÍ, AND M. GERTLER (1998): "Monetary Policy Rules in Practice: Some International Evidence." *European Economic Review* 42, pp. 1033–1067.

- CLARIDA, R., J. GALÍ, AND M. GERTLER (1998): "Monetary Policy Rules and Macroeconomic Stability: Evidence and Some Theory." *The Quarterly Journal of Economics* 115, pp. 147–180.
- CÚRDIA, V. AND M. WOODFORD (2010): "Credit Spreads and Monetary Policy." *Journal of Money, Credit and Banking* 42(s1), pp. 3–35.
- DEVEREUX, M. AND A. SUTHERLAND (2007): "Country Portfolios in Open Economy Macro Models." *Journal of the European Economic Association* 5(2–3), pp. 491–499.
- DETKEN, C. AND F. SMETS (2004): "Asset Price Booms and Monetary Policy." ECB Working Paper No. 364.
- DUBOIS, É. AND E. MICHAUX (2009): "Grocer 1.4: An Econometric Toolbox for Scilab", available at http://dubois.ensae.net/grocer.html.
- EICHENGREEN, B. AND M. D. BORDO (2002): "Crises Now and Then: What Lessons from the Last Era of Financial Globalization." NBER Working Paper No. 8716.
- ENGEL, C. AND A. MATSUMOTO (2009): "The International Diversification Puzzle When Prices are Sticky: It's Really about Exchange-Rate Hedging not Equity Portfolios." *American Economic Journal: Macroeconomics* 1(2), pp.155–188.
- FAIA, E. AND T. MONACELLI (2007): "Optimal Monetary Policy Rules, Asset Prices and Credit Frictions." *Journal of Economic Dynamics and Control* 31(10), pp.3228–3254.
- FISHER, I. (1933): "Debt-deflation Theory of Great Depressions." Econometrica 1(4), pp. 337–357.
- FUHRER, J. AND G. TOOTELL (2008): "Eyes on the Prize: How did the Fed Respond to the Stock Market?" *Journal of Monetary Economics* 55(4), pp. 796–805.
- GOLDSTEIN, M., G. L. KAMINSKY, AND C. M. REINHART (2000): "Assessing Financial Vulnerability: An Early Warning System for Emerging Markets." Washington, DC: Institute for International Economics.
- GOODHART, C. (2006): "A Framework for Assessing Financial Stability?" *Journal of Banking and Finance* 30(12), pp. 3415–3422.
- GOODHART, C., C. OSORIO, AND D. TSOMOCOS (2009): "An Analysis of Monetary Policy and Financial Stability: A New Paradigm." CESifo Working Paper No. 2885.
- GREENSPAN, A. (2007): *The Age of Turbulence: Adventures in a New World.* Penguin Press: New York.
- GRUEN, D., M. PLUMB, AND A. STONE (2005): "How Should Monetary Policy Respond to Asset-Price Bubbles?" *International Journal of Central Banking*, December, pp. 1–31.
- HAKKIO, C. S. AND W. R. KEETON (2009): "Financial Stress: What Is It, How Can It Be Measured, and Why Does It Matter?" *Economic Review* (2Q 2009), Federal Reserve Bank of Kansas City, 1–50.
- IACOVELLO, M. AND S. NERI (2010): "Housing Market Spillovers: Evidence from an Estimated DSGE Model." *American Economic Journal: Macroeconomics* 2(2), pp. 125–164.

- ILLING, M. AND Y. LIU (2006): "Measuring Financial Stress in a Developed Country: An Application to Canada." *Journal of Financial Stability* 2(3), pp. 243–265.
- JENNERGREN, P. (2002): "The Swedish Finance Company Crisis Could It Have Been Anticipated?" Scandinavian Economic History Review 50(2), pp. 7–30.
- JONUNG, L. (2009): "Financial Crisis and Crisis Management in Sweden. Lessons for Today." Asian Development Bank Institute Working Paper No. 156.
- KAMINSKY, G. AND C. REINHART (1999): "The Twin Crises: The Causes of Banking and Balanceof-Payments Problems." *American Economic Review* 89(3), pp. 473–500.
- KIM, C.-J. (2006): "Time-Varying Parameter Models with Endogenous Regressors." *Economics Letters* 91, pp. 21–26.
- KIM, C.-J. AND C. R. NELSON (2006): "Estimation of a Forward-Looking Monetary Policy Rule: A Time-Varying Parameter Model Using Ex-post Data." *Journal of Monetary Economics* 53, pp. 1949–1966.
- KIM, C.-J., K. KISHOR AND C. R. NELSON (2006): "A Time-Varying Parameter Model for a Forward-Looking Monetary Policy Rule based on Real-Time Data." Mimeo.
- KUTTNER, K. N. AND A. S. POSEN (1999): "Does Talk Matter After All? Inflation Targeting and Central Bank Behavior." Federal Reserve Bank of New York Staff Report No. 88.
- LAEVEN, L. AND F. VALENCIA (2008): "Systemic Banking Crisis: A New Database." IMF Working Paper No. 08/224.
- MELVIN, M. AND M. P. TAYLOR (2009): "The Crisis in the Foreign Exchange Market." Journal of International Money and Finance 28(8), pp. 1317–1330.
- MERTON, R. (1974): "On the Pricing of Corporate Debt: The Risk Structure of Interest Rates." *Journal of Finance* 29(2), pp. 449–470.
- MISHKIN, F. (2009): "Is Monetary Policy Effective during Financial Crises?" American Economic Review Papers & Proceedings 99(2), pp. 573–577.
- MISHKIN, F. (2010): "Monetary Policy Flexibility, Risk Management and Financial Disruptions." *Journal of Asian Economics* 21(3), pp. 242–246.
- KIYOTAKI, N. AND J. MOORE (1997): "Credit Cycles." *The Journal of Political Economy* 105, pp. 211–248.
- ORPHANIDES, A. (2001): "Monetary Policy Rules Based on Real-Time Data." *American Economic Review* 91, pp. 964–985.
- POSEN, A. (2006): "Why Central Banks Should Not Burst Bubbles." *International Finance* 9(1), pp. 109–124.
- REINHART, C. M. AND K. S. ROGOFF (2008): "This Time is Different: A Panoramic View of Eight Centuries of Financial Crises." NBER Working Paper No. 13882.

- REIS, R. (2010): "Interpreting the Unconventional U.S. Monetary Policy of 2007-09." NBER Working Paper No. 15662.
- RIGOBON, R. AND B. SACK (2003): "Measuring the Reaction of Monetary Policy to the Stock Market." *The Quarterly Journal of Economics* 118(2), pp. 639–669.
- ROUBINI, N. (2006): "Why Central Banks Should Burst Bubbles." *International Finance* 9(1), pp. 87–107.
- RUDEBUSCH, G. (2006): "Monetary Policy Inertia: Fact or Fiction?" International Journal of Central Banking 2(4), pp. 85–136.
- SEKINE, T. AND Y. TERANISHI (2008): "Inflation Targeting and Monetary Policy Activism." IMES Discussion Paper Series 2008-E-13, Bank of Japan.
- SIKLOS, P. AND M. BOHL (2008): "Asset Prices as Indicators of Euro Area Monetary Policy: An Empirical Assessment of Their Role in a Taylor Rule." *Open Economies Review* 20(1), pp. 39–59.
- SIMS, C. AND T. ZHA (2006): "Were there Regime Switches in US Monetary Policy?" American Economic Review 96(1), pp. 54–81.
- SCHLICHT, E. (1981): "A Seasonal Adjustment Principle and a Seasonal Adjustment Method Derived From this Principle." *Journal of the American Statistical Association* 76(374), pp. 374–378.
- SCHLICHT, E. (2005): "Estimating the Smoothing Parameter in the So-called Hodrick-Prescott Filter." *Journal of the Japan Statistical Society* 35(1), pp. 99–119.
- SCHLICHT, E. AND J. LUDSTECK (2006): "Variance Estimation in a Random Coefficients Model." IZA Discussion Paper No. 2031.
- STOCK, J. H. AND M. W. WATSON (1998): "Median Unbiased Estimation of Coefficient Variance in a Time-Varying Parameter Model." *Journal of the American Statistical Association* 93(441), pp. 349–358.
- TAYLOR, J. B. (1993): "Discretion versus Policy Rules in Practice." *Carnegie-Rochester Conference* Series on Public Policy 39, pp. 195–214.
- TAYLOR, J. B. (ED.), 1999. Monetary Policy Rules. The University of Chicago Press: Chicago.
- TAYLOR, J. B. (2001): "The Role of the Exchange Rate in Monetary Policy Rules." American Economic Review 91(2), pp. 263–267.
- TAYLOR, J. B. (2008): "Monetary Policy and the State of the Economy." Testimony before the Committee on Financial Services, U.S. House of Representatives, February 26, 2008.
- TAYLOR, J. B. (2010): "Getting Back on Track: Macroeconomic Policy Lessons from the Financial Crisis." *Federal Reserve Bank of St. Louis Review* 92(3) (May/June 2010), pp. 165–176.
- TAYLOR, J. B. AND J. WILLIAMS (2009): "A Black Swan in the Money Market." American Economic Journal: Macroeconomics 1(1), pp. 58–83.

- TAYLOR, M. P. AND E. DAVRADAKIS (2006): "Interest Rate Setting and Inflation Targeting: Evidence of a Nonlinear Taylor Rule for the United Kingdom." *Studies in Nonlinear Dynamics & Econometrics* 10(4), Article 1.
- TERANISHI, Y. (2009): "Credit Spread and Monetary Policy." IMES Discussion Paper Series 2009-E-14, Bank of Japan.
- TERRONES, M. AND C. OTROK (2004): "The Global House Price Boom." In: IMF: World Economic Outlook, Chapter 3, April. Washington: International Monetary Fund.
- TRECROCI, C. AND M. VASSALLI (2010): "Monetary Policy Regime Shifts: New Evidence from Time-Varying Interest-Rate Rules." *Economic Inquiry* 48(4), pp. 933–950.
- TOVAR, C. E. (2009): "DSGE Models and Central Banks." *Economics: The Open-Access, Open-*Assessment E-Journal 3, 200916.
- VALENTE, G. (2003): "Monetary Policy Rules and Regime Shift." *Applied Financial Economics* 13, pp. 525–535.
- ZHANG, C., D. R. OSBORN, AND H. DONG (2008): "The New Keynesian Phillips Curve: From Sticky Inflation to Sticky Prices." *Journal of Money, Credit and Banking* 40, pp. 667–699.

Appendix 1: Time-Varying Monetary Policy Rule Estimates



Figure A.1.1: Time-Varying Monetary Policy Rules: USA

Note: The estimated coefficients of the time-varying monetary policy rule are depicted with a 95% confidence interval.



Figure A.1.2: Time-Varying Monetary Policy Rules: UK

Note: The estimated coefficients of the time-varying monetary policy rule are depicted with a 95% confidence interval.



Figure A.1.3: Time-Varying Monetary Policy Rules: Sweden

Note: The estimated coefficients of the time-varying monetary policy rule are depicted with a 95% confidence interval.



Figure A.1.4: Time-Varying Monetary Policy Rules: Australia

Note: The estimated coefficients of the time-varying monetary policy rule are depicted with a 95% confidence interval.



Figure A.1.5: Time-Varying Monetary Policy Rules: Canada

Note: The estimated coefficients of the time-varying monetary policy rule are depicted with a 95% confidence interval.

Appendix 2: The Results with the Interbank Rate as the Dependent Variable in the Policy Rule



Figure A2.1: The Effect of Financial Stress on Interest-Rate Setting

Notes: The figure depicts the evolution of the financial-stress effect. The stress effect (y-axis) is defined as the product of the estimated coefficient on the financial-stress indicator in the monetary-policy rule and the value of the IMF financial-stress indicator (δx). The stress effect shows the magnitude of the interest-rate reaction to financial stress in percentage points.



Figure A2.2: The Effect of Financial Stress Components on Interest-Rate Setting: Bank Stress, Exchange-Rate Stress and Stock-Market Stress

Notes: The figure depicts the evolution of the components of the financial-stress effect, namely, the bank stress effect, the exchange-rate stress effect, and the stock-market stress effect. The stress effect (y-axis) is defined as the product of the estimated coefficient on the given component of the financial-stress indicator in the monetary-policy rule and the value of the corresponding component of the IMF financial-stress indicator (δx). The stress effect shows the magnitude of the interest-rate reaction to financial stress in percentage points.

Appendix 3: The Results with Different Leads and Lag of the FSI

Figure A3.1: The Effect of Financial Stress (t-1 vs. t-2, t, t+1, t+2) on Interest-Rate Setting



Appendix 4: The Results with Individual Variables of Bank Stress and Stock-Market Stress



Figure A4.1: The Effect of Bank Stress on Interest-Rate Setting





Figure A4.2: The Effect of Stock-Market Stress on Interest-Rate Setting

Appendix 5: Significance of the Financial Stress Index in the Estimated Taylor Rules

	α	β	γ	ρ	δ	J-statistics	p-value
				United Stat	es		
1	5.59	1.59	1.51	0.97	-0.014	24.9808	0.6289
	(1.42)	(0.99)	(0.53)	(0.01)	(0.006)		
2	-9.42	0.45	0.94	0.94		15.0451	0.8207
	(4.78)	(0.33)	(0.34)	(0.01)			
			U	nited Kingo	lom		
1	3.93	0.37	1.51	0.97	-0.018	15.0423	0.5212
	(1.31)	(0.52)	(0.39)	(0.01)	(0.004)		
2	7.68	-0.89	2.77	0.98		11.4534	0.4905
	(2.78)	(0.74)	(0.71)	(0.01)			
				Sweden			
1	-1.87	2.59	-0.16	0.84	-0.135	24.9808	0.6289
	(0.86)	(0.46)	(0.22)	(0.04)	(0.029)		
2	0.24	2.03	-0.12	0.76		15.0451	0.8207
	(0.53)	(0.39)	(0.16)	(0.05)			
				Australia			
1	0.04	2.06	-0.02	0.95	-0.038	21.5261	0.9731
	(0.79)	(0.3)	(0.1)	(0.01)	(0.006)		
2	2.2	1.84	0.19	0.89		15.2464	0.9830
	(0.93)	(0.22)	(0.14)	(0.02)			
				Canada			
1	-0.33	2.07	0.87	0.89	-0.089	10.6859	0.8284
	(1.23)	(0.96)	(0.30)	(0.04)	(0.023)		
2	1.21	1.67	0.75	0.86		9.4332	0.7395
	(1.23)	(0.87)	(0.28)	(0.05)			
			United Stat	es: 1981:1–1	1999:12 sample	e	
1	1.82	1.43	1.48	0.95	-0.015	20.1672	0.8583
	(1.27)	(0.57)	(0.27)	(0.01)	(0.007)	10.0017	0.040.0
1*	-0.25	2.18	0.3	(0.8')	-0.043	19.8946	0.8683
	(0.//)	(U.42)	(0.00)	(0.02)	1082.10 100	4.12 as much	
		United Sta	ites: Clarida	et al. (1998) ·	- 1982:10-199	74:12 sample	
2	-0.1	1.83	0.56	0.97		10.9000	0.9980
	(1.54)	(0.45)	(0.16)	(0.03)			

Table A5.1: Time-Invariant Reaction Functions, GMM estimates

Notes: Numbers in (.) are standard errors. The samples are as follows: United States: 1981:1M–2009:6M, United Kingdom: 1981:1M–2009:3M, Australia: 1983:3M–2009:5M, Sweden 1984:2Q–20091Q, Canada 1981:1Q–2008:4Q. Model 1: $rt = (1-\rho)(\alpha + \beta \pi t + k + \gamma yt) + \rho rt - 1 + \delta xt - 1$. Model 2 does not contain financial stress: $rt = (1-\rho)(\alpha + \beta \pi t + k + \gamma yt) + \rho rt - 1$. k equals 6 for the USA, the UK and Australia, 2 for Sweden and 4 for Canada. For Sweden, a dummy variable for the third quarter of 1992 (the RM crisis) is included.

Both models are estimated using the GMM. The list of instruments follows. United States: lags of interest rate, output gap, inflation, and financial stress (1, 2, 3, 4, 5, 6, 9, 12), model 2 without lags of FSI in a set of instruments. United Kingdom: lags of interest rate, output gap, inflation, and EURIBOR 3M (1, 2, 3, 4, 5, 6, 9, 12), FSI (1,2,3). Australia: interest rate, inflation, output gap, US money market rate and FSI (1, 2, 3, 4, 5, 6, 9, 12). Sweden: lags of interest rate, inflation, output gap, EURIBOR 3M, and FSI (1, 2, 3, 4) + the dummy for the ERM crisis. Canada: interest rate, inflation, output gap, U.S. money market rate, and FSI (1, 2, 3, 4) + the dummy for the ERM crisis. Canada: interest rate, inflation, output gap, U.S. money market rate, and FSI (1, 2, 3, 4).

Additionally, we show the results for the USA estimated on the subsample 1981–1999. The model denoted as 1* has the output gap derived from the quadratic trend of log industrial production in a similar fashion as in Clarida et al. (1998). Their results are provided for comparison with ours.

Table A5.2: Periods with Significant Responses to Financial Stress

		1980s	1990s	2000s
United States	2SD			2008M03-2009M03
	1SD	1982M11-	-1992M09	2007M05-2009M06
United				
Kingdom	2SD	1987M08-1989M11		2007M09-2009M03
	1SD	1987M01-	-1993M01	2006M03-2009M03
Sweden	2SD			
	1SD		1990Q2-1992Q2	2001Q2-2002Q3
			1993Q1	2009Q1
			1999Q4-2000Q2	
Australia	2SD	1987M04-1988M10		2008M09-2009M03
	1SD	1983M07-	-1993M10	2002M10-2009M05
			1996M06-1997M05	
Canada	2SD			_
	1SD	1982Q3-1984Q1	1992Q3-1995Q4	-

CNB WORKING PAPER SERIES

3/2011	Jaromír Baxa Roman Horváth Bořek Vašíček	Time-varying monetary-policy rules and financial stress: Does financial instability matter for monetary policy?
2/2011	Marek Rusnák Tomáš Havránek Roman Horváth	How to solve the price puzzle? A Meta-analysis
1/2011	Jan Babecký Aleš Bulíř Kateřina Šmídková	Sustainable real exchange rates in the new EU member states: What did the I reat Tecession change?
15/2010	Ke Pang Pierre L. Siklos	Financial frictions and credit spreads
14/2010	Filip Novotný Marie Raková	Assessment of consensus forecasts accuracy: The Czech National Bank perspective
13/2010	Jan Filáček Branislav Saxa	Central bank forecasts as a coordination device
12/2010	Kateřina Arnoštová David Havrlant Luboš Růžička Peter Tóth	Short-term forecasting of Czech quarterly GDP using monthly indicators
11/2010	Roman Horváth Kateřina Šmídková Jan Zápal	Central banks' voting records and future policy
10/2010	Alena Bičáková Zuzana Prelcová Renata Pašaličová	Who borrows and who may not repay?
9/2010	Luboš Komárek Jan Babecký Zlatuše Komárková	Financial integration at times of financial instability
8/2010	Kamil Dybczak Peter Tóth David Voňka	Effects of price shocks to consumer demand. Estimating the QUAIDS demand system on Czech Household Budget Survey data
7/2010	Jan Babecký Philip Du Caju Theodora Kosma Martina Lawless Julián Messina Tairi Rõõm	The margins of labour cost adjustment: Survey evidence from European Firms
6/2010	Tomáš Havránek Roman Horváth Jakub Matějů	Do financial variables help predict macroeconomic environment? The case of the Czech Republic
5/2010	Roman Horváth Luboš Komárek Filip Rozsypal	Does money help predict inflation? An empirical assessment for Central Europe
4/2010	Oxana Babecká Kucharčuková Jan Babecký Martin Raiser	A Gravity approach to modelling international trade in South- Eastern Europe and the Commonwealth of Independent States: The role of geography, policy and institutions
3/2010	Tomáš Havránek Zuzana Iršová	Which foreigners are worth wooing? A Meta-analysis of vertical spillovers from FDI

2/2010	Jaromír Baxa Roman Horváth Bořek Vašíček	<i>How does monetary policy change? Evidence on inflation targeting countries</i>
1/2010	Adam Geršl Petr Jakubík	Relationship lending in the Czech Republic
15/2009	David N. DeJong Roman Liesenfeld Guilherme V. Moura Jean-Francois Richard Hariharan Dharmarajan	Efficient likelihood evaluation of state-space representations
14/2009	Charles W. Calomiris	Banking crises and the rules of the game
13/2009	Jakub Seidler Petr Jakubík	The Merton approach to estimating loss given default: Application to the Czech Republic
12/2009	Michal Hlaváček Luboš Komárek	Housing price bubbles and their determinants in the Czech Republic and its regions
11/2009	Kamil Dybczak Kamil Galuščák	Changes in the Czech wage structure: Does immigration matter?
10/2009	Jiří Böhm Petr Král Branislav Saxa	Percepion is always right: The CNB's monetary policy in the media
9/2009	Alexis Derviz Marie Raková	Funding costs and loan pricing by multinational bank affiliates
8/2009	Roman Horváth Anca Maria Podpiera	Heterogeneity in bank pricing policies: The Czech evidence
7/2009	David Kocourek Filip Pertold	The impact of early retirement incentives on labour market participation: Evidence from a parametric change in the Czech Republic
6/2009	Nauro F. Campos Roman Horváth	Reform redux: Measurement, determinants and reversals
5/2009	Kamil Galuščák Mary Keeney Daphne Nicolitsas Frank Smets Pawel Strzelecki Matija Vodopivec	The determination of wages of newly hired employees: Survey evidence on internal versus external factors
4/2009	Jan Babecký Philip Du Caju Theodora Kosma Martina Lawless Julián Messina Tairi Rõõm	Downward nominal and real wage rigidity: Survey evidence from European firms
3/2009	Jiri Podpiera Laurent Weill	Measuring excessive risk-taking in banking
2/2009	Michal Andrle Tibor Hlédik Ondra Kameník Jan Vlček	Implementing the new structural model of the Czech National Bank
1/2000	Kamil Dash anala	The impact of population agains on the Czech economy

	Jan Babecký	
14/2008	Gabriel Fagan Vitor Gaspar	Macroeconomic adjustment to monetary union
13/2008	Giuseppe Bertola Anna Lo Prete	<i>Openness, financial markets, and policies: Cross-country and dynamic patterns</i>
12/2008	Jan Babecký Kamil Dybczak Kamil Galuščák	Survey on wage and price formation of Czech firms
11/2008	Dana Hájková	The measurement of capital services in the Czech Republic
10/2008	Michal Franta	<i>Time aggregation bias in discrete time models of aggregate duration data</i>
9/2008	Petr Jakubík Christian Schmieder	Stress testing credit risk: Is the Czech Republic different from Germany?
8/2008	Sofia Bauducco Aleš Bulíř	Monetary policy rules with financial instability
	Martin Cihák	
7/2008	Jan Brůha Jiří Podpiera	The origins of global imbalances
6/2008	Jiří Podpiera Marie Raková	The price effects of an emerging retail market
5/2008	Kamil Dybczak David Voňka Nico van der Windt	The effect of oil price shocks on the Czech economy
4/2008	Magdalena M. Borys Roman Horváth	The effects of monetary policy in the Czech Republic: An empirical study
3/2008	Martin Cincibuch Tomáš Holub Jaromír Hurník	Central bank losses and economic convergence
2/2008	Jiří Podpiera	Policy rate decisions and unbiased parameter estimation in conventionally estimated monetary policy rules
1/2008	Balázs Égert Doubravko Mihaljek	Determinants of house prices in Central and Eastern Europe
17/2007	Pedro Portugal	U.S. unemployment duration: Has long become longer or short become shorter?
16/2007	Yuliya Rychalovská	Welfare-based optimal monetary policy in a two-sector small open economy
15/2007	Juraj Antal František Brázdik	The effects of anticipated future change in the monetary policy regime
14/2007	Aleš Bulíř Kateřina Šmídková Viktor Kotlán David Navrátil	Inflation targeting and communication: Should the public read inflation reports or tea leaves?
13/2007	Martin Cinncibuch Martina Horníková	<i>Measuring the financial markets' perception of EMU enlargement:</i> <i>The role of ambiguity aversion</i>
12/2007	Oxana Babetskaia- Kukharchuk	Transmission of exchange rate shocks into domestic inflation: The case of the Czech Republic
11/2007	Jan Filáček	Why and how to assess inflation target fulfilment
10/2007	Michal Franta Branislav Saxa	Inflation persistence in new EU member states: Is it different than in the Euro area members?

0/2007	Katerina Smidkova	Un sure la sure d'in a divide de sure in de s Cara de Darre bliss
9/2007	Jan Pavel	Incentive effects of policies
8/2007	Adam Geršl Ieva Rubene Tina Zumer	Foreign direct investment and productivity spillovers: Updated evidence from Central and Eastern Europe
7/2007	Ian Babetskii Luboš Komárek Zlatuše Komárková	Financial integration of stock markets among new EU member states and the euro area
6/2007	Anca Pruteanu-Podpiera Laurent Weill Franziska Schobert	Market power and efficiency in the Czech banking sector
5/2007	Jiří Podpiera Laurent Weill	Bad luck or bad management? Emerging banking market experience
4/2007	Roman Horváth	<i>The time-varying policy neutral rate in real time: A predictor for future inflation?</i>
3/2007	Jan Brůha Jiří Podpiera Stanislav Polák	The convergence of a transition economy: The case of the Czech Republic
2/2007	Ian Babetskii Nauro F. Campos	Does reform work? An econometric examination of the reform-growth puzzle
1/2007	Ian Babetskii Fabrizio Coricelli Roman Horváth	Measuring and explaining inflation persistence: Disaggregate evidence on the Czech Republic
13/2006	Frederic S. Mishkin Klaus Schmidt- Hebbel	Does inflation targeting make a difference?
12/2006	Richard Disney Sarah Bridges John Gathergood	Housing wealth and household indebtedness: Is there a household 'financial accelerator'?
11/2006	Michel Juillard Ondřej Kameník Michael Kumhof Douglas Laxton	Measures of potential output from an estimated DSGE model of the United States
10/2006	Jiří Podpiera Marie Raková	Degree of competition and export-production relative prices when the exchange rate changes: Evidence from a panel of Czech exporting companies
9/2006	Alexis Derviz Jiří Podpiera	Cross-border lending contagion in multinational banks
8/2006	Aleš Bulíř Jaromír Hurník	The Maastricht inflation criterion: "Saints" and "Sinners"
7/2006	Alena Bičáková Jiří Slačálek Michal Slavík	Fiscal implications of personal tax adjustments in the Czech Republic
6/2006	Martin Fukač Adrian Pagan	Issues in adopting DSGE models for use in the policy process
5/2006	Martin Fukač	New Keynesian model dynamics under heterogeneous expectations and adaptive learning
4/2006	Kamil Dybczak	Supply-side performance and structure in the Czech Republic

	Vladislav Flek Dana Hájková Jaromír Hurník	(1995–2005)
3/2006	Aleš Krejdl	Fiscal sustainability – definition, indicators and assessment of Czech public finance sustainability
2/2006	Kamil Dybczak	Generational accounts in the Czech Republic
1/2006	Ian Babetskii	Aggregate wage flexibility in selected new EU member states
14/2005	Stephen G. Cecchetti	The brave new world of central banking: The policy challenges posed by asset price booms and busts
13/2005	Robert F. Engle Jose Gonzalo Rangel	The spline GARCH model for unconditional volatility and its global macroeconomic causes
12/2005	Jaromír Beneš Tibor Hlédik Michael Kumhof David Vávra	An economy in transition and DSGE: What the Czech national bank's new projection model needs
11/2005	Marek Hlaváček Michael Koňák Josef Čada	The application of structured feedforward neural networks to the modelling of daily series of currency in circulation
10/2005	Ondřej Kameník	Solving SDGE models: A new algorithm for the sylvester equation
9/2005	Roman Šustek	Plant-level nonconvexities and the monetary transmission mechanism
8/2005	Roman Horváth	Exchange rate variability, pressures and optimum currency area criteria: Implications for the central and eastern european countries
7/2005	Balázs Égert Luboš Komárek	Foreign exchange interventions and interest rate policy in the Czech Republic: Hand in glove?
6/2005	Anca Podpiera Jiří Podpiera	Deteriorating cost efficiency in commercial banks signals an increasing risk of failure
5/2005	Luboš Komárek Martin Melecký	The behavioural equilibrium exchange rate of the Czech koruna
4/2005	Kateřina Arnoštová Jaromír Hurník	The monetary transmission mechanism in the Czech Republic (evidence from VAR analysis)
3/2005	Vladimír Benáček Jiří Podpiera Ladislav Prokon	Determining factors of Czech foreign trade: A cross-section time series perspective
2/2005	Kamil Galuščák Daniel Münich	Structural and cyclical unemployment: What can we derive from the matching function?
1/2005	Ivan Babouček Martin Jančar	Effects of macroeconomic shocks to the quality of the aggregate loan portfolio
10/2004	Aleš Bulíř Kateřina Šmídková	Exchange rates in the new EU accession countries: What have we learned from the forerunners
9/2004	Martin Cincibuch Jiří Podpiera	Beyond Balassa-Samuelson: Real appreciation in tradables in transition countries
8/2004	Jaromír Beneš David Vávra	<i>Eigenvalue decomposition of time series with application to the Czech business cycle</i>
7/2004	Vladislav Flek, ed.	Anatomy of the Czech labour market: From over-employment to

under-employment in ten years?

6/2004	Narcisa Kadlčáková Joerg Keplinger	Credit risk and bank lending in the Czech Republic
5/2004	Petr Král	Identification and measurement of relationships concerning inflow of FDI: The case of the Czech Republic
4/2004	Jiří Podpiera	Consumers, consumer prices and the Czech business cycle identification
3/2004	Anca Pruteanu	The role of banks in the Czech monetary policy transmission mechanism
2/2004	Ian Babetskii	EU enlargement and endogeneity of some OCA criteria: Evidence from the CEECs
1/2004	Alexis Derviz Jiří Podpiera	Predicting bank CAMELS and S&P ratings: The case of the Czech Republic

CNB RESEARCH AND POLICY NOTES				
1/2008	Nicos Christodoulakis	<i>Ten years of EMU: Convergence, divergence and new policy prioritie</i>		
2/2007	Carl E. Walsh	Inflation targeting and the role of real objectives		
1/2007	Vojtěch Benda Luboš Růžička	Short-term forecasting methods based on the LEI approach: The case of the Czech Republic		
2/2006	Garry J. Schinasi	Private finance and public policy		
1/2006	Ondřej Schneider	The EU budget dispute – A blessing in disguise?		
5/2005	Jan Stráský	<i>Optimal forward-looking policy rules in the quarterly projection model of the Czech National Bank</i>		
4/2005	Vít Bárta	Fulfilment of the Maastricht inflation criterion by the Czech Republic: Potential costs and policy options		
3/2005	Helena Sůvová Eva Kozelková David Zeman Jaroslava Bauerová	Eligibility of external credit assessment institutions		
2/2005	Martin Čihák Jaroslav Heřmánek	Stress testing the Czech banking system: Where are we? Where are we going?		
1/2005	David Navrátil Viktor Kotlán	The CNB's policy decisions – Are they priced in by the markets?		
4/2004	Aleš Bulíř	External and fiscal sustainability of the Czech economy: A quick look through the IMF's night-vision goggles		
3/2004	Martin Čihák	Designing stress tests for the Czech banking system		
2/2004	Martin Čihák	Stress testing: A review of key concepts		
1/2004	Tomáš Holub	Foreign exchange interventions under inflation targeting: The Czech experience		

CNB ECONOMIC RESEARCH BULLETIN

April 2011	Monetary policy analysis in a central bank
November 2010	Wage adjustment in Europe
May 2010	Ten years of economic research in the CNB
November 2009	Financial and global stability issues
May 2009	Evaluation of the fulfilment of the CNB's inflation targets 1998–2007
December 2008	Inflation targeting and DSGE models
April 2008	Ten years of inflation targeting
December 2007	Fiscal policy and its sustainability
August 2007	Financial stability in a transforming economy
November 2006	ERM II and euro adoption
August 2006	Research priorities and central banks
November 2005	Financial stability
May 2005	Potential output
October 2004	Fiscal issues
May 2004	Inflation targeting
December 2003	Equilibrium exchange rate

Czech National Bank Economic Research Department Na Příkopě 28, 115 03 Praha 1 Czech Republic phone: +420 2 244 12 321 fax: +420 2 244 14 278 http://www.cnb.cz e-mail: research@cnb.cz ISSN 1803-7070